Maps of Bounded Rationality: Psychology for Behavioral Economics[†]

By Daniel Kahneman*

The work cited by the Nobel committee was done jointly with Amos Tversky (1937–1996) during a long and unusually close collaboration. Together, we explored the psychology of intuitive beliefs and choices and examined their bounded rationality. Herbert A. Simon (1955, 1979) had proposed much earlier that decision makers should be viewed as boundedly rational, and had offered a model in which utility maximization was replaced by satisficing. Our research attempted to obtain a map of bounded rationality, by exploring the systematic biases that separate the beliefs that people have and the choices they make from the optimal beliefs and choices assumed in rational-agent models. The rational-agent model was our starting point and the main source of our null hypotheses, but Tversky and I viewed our research primarily as a contribution to psychology, with a possible contribution to economics as a secondary benefit. We were drawn into the interdisciplinary conversation by economists who hoped that psychology could be a useful source of assumptions for economic theorizing, and indirectly a source of hypotheses for economic research (Richard H. Thaler, 1980, 1991, 1992). These

hopes have been realized to some extent, giving rise to an active program of research by behavioral economists (Thaler, 2000; Colin Camerer et al., forthcoming; for other examples, see Kahneman and Tversky, 2000).

My work with Tversky comprised three separate programs of research, some aspects of which were carried out with other collaborators. The first explored the heuristics that people use and the biases to which they are prone in various tasks of judgment under uncertainty, including predictions and evaluations of evidence (Kahneman and Tversky, 1973; Tversky and Kahneman, 1974; Kahneman et al., 1982). The second was concerned with prospect theory, a model of choice under risk (Kahneman and Tversky, 1979; Tversky and Kahneman, 1992) and with loss aversion in riskless choice (Kahneman et al., 1990, 1991; Tversky and Kahneman, 1991). The third line of research dealt with framing effects and with their implications for rational-agent models (Tversky and Kahneman, 1981, 1986). The present essay revisits these three lines of research in light of recent advances in the psychology of intuitive judgment and choice. Many of the ideas presented here were anticipated informally decades ago, but the attempt to integrate them into a coherent approach to judgment and choice is recent.

Economists often criticize psychological research for its propensity to generate lists of errors and biases, and for its failure to offer a coherent alternative to the rational-agent model. This complaint is only partly justified: psychological theories of intuitive thinking cannot match the elegance and precision of formal normative models of belief and choice, but this is just another way of saying that rational models are psychologically unrealistic. Furthermore, the alternative to simple and precise models is not chaos. Psychology offers integrative concepts and mid-level generalizations, which gain credibility from their ability to explain ostensibly different phenomena in diverse domains. In this spirit, the present essay offers a unified

[†] This article is a revised version of the lecture Daniel Kahneman delivered in Stockholm, Sweden, on December 8, 2002, when he received the Bank of Sweden Prize in Economic Sciences in Memory of Alfred Nobel. The article is copyright © The Nobel Foundation 2002 and is published here with the permission of the Nobel Foundation.

^{*} Woodrow Wilson School, Princeton University, Princeton, NJ 08544 (e-mail: Kahneman@princeton.edu). This essay revisits problems that Amos Tversky and I studied together many years ago, and continued to discuss in a conversation that spanned several decades. It builds on an analysis of judgment heuristics that was developed in collaboration with Shane Frederick (Kahneman and Frederick, 2002). A different version was published in *American Psychologist* in September 2003. For detailed comments on this version I am grateful to Angus Deaton, David Laibson, Michael Rothschild, and Richard Thaler. The usual caveats apply. Geoffrey Goodwin, Amir Goren, and Kurt Schoppe provided helpful research assistance.

treatment of intuitive judgment and choice, which builds on an earlier study of the relationship between preferences and attitudes (Kahneman et al., 1999) and extends a model of judgment heuristics recently proposed by Kahneman and Shane Frederick (2002). The guiding ideas are (i) that most judgments and most choices are made intuitively; (ii) that the rules that govern intuition are generally similar to the rules of perception. Accordingly, the discussion of the rules of intuitive judgments and choices will rely extensively on visual analogies.

Section I introduces a distinction between two generic modes of cognitive function, corresponding roughly to intuition and reasoning. Section II describes the factors that determine the relative accessibility of different judgments and responses. Section III relates prospect theory to the general proposition that changes and differences are more accessible than absolute values. Section IV explains framing effects in terms of differential salience and accessibility. Section V reviews an attribute substitution model of heuristic judgment. Section VI describes a particular family of heuristics, called prototype heuristics. Section VII discusses the interactions between intuitive and deliberate thought. Section VIII concludes.

I. The Architecture of Cognition: Two Systems

The present treatment distinguishes two modes of thinking and deciding, which correspond roughly to the everyday concepts of reasoning and intuition. Reasoning is what we do when we compute the product of 17 by 258, fill an income tax form, or consult a map. Intuition is at work when we read the sentence "Bill Clinton is a shy man" as mildly amusing, or when we find ourselves reluctant to eat a piece of what we know to be chocolate that has been formed in the shape of a cockroach (Paul Rozin and Carol Nemeroff, 2002). Reasoning is done deliberately and effortfully, but intuitive thoughts seem to come spontaneously to mind, without conscious search or computation, and without effort. Casual observation and systematic research indicate that most thoughts and actions are normally intuitive in this sense (Daniel T. Gilbert, 1989, 2002; Timothy D. Wilson, 2002; Seymour Epstein, 2003).

Although effortless thought is the norm, some monitoring of the quality of mental oper-

ations and overt behavior also goes on. We do not express every passing thought or act on every impulse. But the monitoring is normally lax, and allows many intuitive judgments to be expressed, including some that are erroneous (Kahneman and Frederick, 2002). Ellen J. Langer et al. (1978) provided a well-known example of what she called "mindless behavior." In her experiment, a confederate tried to cut in line at a copying machine, using various preset "excuses." The conclusion was that statements that had the form of an unqualified request were rejected (e.g., "Excuse me, may I use the Xerox machine?"), but almost any statement that had the general form of an explanation was accepted, including "Excuse me, may I use the Xerox machine because I want to make copies?" The superficiality is striking.

Frederick (2003, personal communication) has used simple puzzles to study cognitive selfmonitoring, as in the following example: "A bat and a ball cost \$1.10 in total. The bat costs \$1 more than the ball. How much does the ball cost?" Almost everyone reports an initial tendency to answer "10 cents" because the sum \$1.10 separates naturally into \$1 and 10 cents, and 10 cents is about the right magnitude. Frederick found that many intelligent people yield to this immediate impulse: 50 percent (47/93) of a group of Princeton students and 56 percent (164/293) of students at the University of Michigan gave the wrong answer. Clearly, these respondents offered their response without first checking it. The surprisingly high rate of errors in this easy problem illustrates how lightly the output of effortless associative thinking is monitored: people are not accustomed to thinking hard, and are often content to trust a plausible judgment that quickly comes to mind. Remarkably, Frederick has found that errors in this puzzle and in others of the same type were significant predictors of high discount rates.

In the examples discussed so far, intuition was associated with poor performance, but intuitive thinking can also be powerful and accurate. High skill is acquired by prolonged practice, and the performance of skills is rapid and effortless. The proverbial master chess player who walks past a game and declares "white mates in three" without slowing is performing intuitively (Simon and William G. Chase, 1973), as is the experienced nurse who



FIGURE 1. THREE COGNITIVE SYSTEMS

detects subtle signs of impending heart failure (Gary Klein, 1998; Atul Gawande, 2002).

The distinction between intuition and reasoning has recently been a topic of considerable interest to psychologists (see, e.g., Shelly Chaiken and Yaacov Trope, 1999; Gilbert, 2002; Steven A. Sloman, 2002; Keith E. Stanovich and Richard F. West, 2002). There is substantial agreement on the characteristics that distinguish the two types of cognitive processes, for which Stanovich and West (2000) proposed the neutral labels of System 1 and System 2. The scheme shown in Figure 1 summarizes these characteristics. The operations of System 1 are fast, automatic, effortless, associative, and often emotionally charged; they are also governed by habit, and are therefore difficult to control or modify. The operations of System 2 are slower, serial, effortful, and deliberately controlled; they are also relatively flexible and potentially rule-governed.

The difference in effort provides the most useful indications of whether a given mental process should be assigned to System 1 or System 2. Because the overall capacity for mental effort is limited, effortful processes tend to disrupt each other, whereas effortless processes neither cause nor suffer much interference when combined with other tasks. For example, a driver's ability to conduct a conversation is a sensitive indicator of the amount of attention currently demanded by the driving task. Dual tasks have been used in hundreds of psychological experiments to measure the attentional demands of different mental activities (for a review, see Harold E. Pashler, 1998). Studies using the dual-task method suggest that the selfmonitoring function belongs with the effortful operations of System 2. People who are occupied by a demanding mental activity (e.g., attempting to hold in mind several digits) are much more likely to respond to another task by blurting out whatever comes to mind (Gilbert, 1989). The phrase that "System 2 monitors the activities of System 1" will be used here as shorthand for a hypothesis about what would happen if the operations of System 2 were disrupted. For example, it is safe to predict that the percentage of errors in the bat-and-ball question will increase, if the respondents are asked this question while attempting to keep a list of words in their active memory.

In the language that will be used here, the perceptual system and the intuitive operations

of System 1 generate *impressions* of the attributes of objects of perception and thought. These impressions are not voluntary and need not be verbally explicit. In contrast, *judgments* are always explicit and intentional, whether or not they are overtly expressed. Thus, System 2 is involved in all judgments, whether they originate in impressions or in deliberate reasoning. The label "intuitive" is applied to judgments that directly reflect impressions.

Figure 1 illustrates an idea that guided the research that Tversky and I conducted from its early days: that intuitive judgments occupy a position-perhaps corresponding to evolutionary history-between the automatic operations of perception and the deliberate operations of reasoning. All the characteristics that students of intuition have attributed to System 1 are also properties of perceptual operations. Unlike perception, however, the operations of System 1 are not restricted to the processing of current stimulation. Like System 2, the operations of System 1 deal with stored concepts as well as with percepts, and can be evoked by language. This view of intuition suggests that the vast store of scientific knowledge available about perceptual phenomena can be a source of useful hypotheses about the workings of intuition. The strategy of drawing on analogies from perception is applied in the following section.

II. The Accessibility Dimension

A defining property of intuitive thoughts is that they come to mind spontaneously, like percepts. The technical term for the ease with which mental contents come to mind is *accessibility* (E. Tory Higgins, 1996). To understand intuition, we must understand why some thoughts are accessible and others are not. The remainder of this section introduces the concept of accessibility by examples drawn from visual perception.

Consider Figures 2a and 2b. As we look at the object in Figure 2a, we have immediate impressions of the height of the tower, the area of the top block, and perhaps the volume of the tower. Translating these impressions into units of height or volume requires a deliberate operation, but the impressions themselves are highly accessible. For other attributes, no perceptual impression exists. For example, the total area that the blocks would cover if the tower were



FIGURE 2. EXAMPLES OF DIFFERENTIAL ACCESSIBILITY

dismantled is not perceptually accessible, though it can be estimated by a deliberate procedure, such as multiplying the area of a block by the number of blocks. Of course, the situation is reversed with Figure 2b. Now the blocks are laid out and an impression of total area is immediately accessible, but the height of the tower that could be constructed with these blocks is not.

Some relational properties are accessible. Thus, it is obvious at a glance that Figures 2a and 2c are different, but also that they are more similar to each other than either is to Figure 2b. And some statistical properties of ensembles are accessible, while others are not. For an example, consider the question "What is the average length of the lines in Figure 3?" This question is easy. When a set of objects of the same general kind is presented to an observerwhether simultaneously or successively-a representation of the set is computed automatically, which includes quite precise information about the average (Dan Ariely, 2001; Sang-Chul Chong and Anne Treisman, 2003). The representation of the prototype is highly accessible, and it has the character of a percept: we form an impression of the typical line without choosing to do so. The only role for System 2 in this task is to map the impression of typical length onto the appropriate scale. In contrast, the answer to the question "What is the total length of the lines in the display?" does not come to mind without considerable effort.

As the example of averages and sums illustrates, some attributes are more accessible than others, both in perception and in judgment. Attributes that are routinely and automatically produced by the perceptual system or by System



FIGURE 3. DIFFERENTIAL ACCESSIBILITY OF STATISTICAL PROPERTIES

1, without intention or effort, have been called *natural assessments* (Tversky and Kahneman, 1983). Kahneman and Frederick (2002) compiled a partial list of these natural assessments. In addition to physical properties such as size, distance, and loudness, the list includes more abstract properties such as similarity, causal propensity, surprisingness, affective valence, and mood.

The evaluation of stimuli as good or bad is a particularly important natural assessment. The evidence, both behavioral (John A. Bargh, 1997; Robert B. Zajonc, 1998) and neurophysiological (e.g., Joseph E. LeDoux, 2000), is consistent with the idea that the assessment of whether objects are good (and should be approached) or bad (should be avoided) is carried out quickly and efficiently by specialized neural circuitry. A remarkable experiment reported by Bargh (1997) illustrates the speed of the evaluation process, and its direct link to approach and avoidance. Participants were shown a series of stimuli on a screen, and instructed to respond to each stimulus as soon as it appeared, by moving a lever that blanked the screen. The stimuli were affectively charged words, some positive (e.g., LOVE) and some aversive (e.g., VOMIT), but this feature was irrelevant to the participant's task. Half the participants responded by pulling the lever toward themselves, half responded by pushing the lever away. Although the response

was initiated within a fraction of a second, well before the meaning of the stimulus was consciously registered, the emotional valence of the word had a substantial effect. Participants were relatively faster in pulling a lever toward themselves (approach) for positive words, and relatively faster pushing the lever away when the word was aversive. The tendencies to approach or avoid were evoked by an automatic process that was not under conscious voluntary control. Several psychologists have commented on the influence of this primordial evaluative system (here included in System 1) on the attitudes and preferences that people adopt consciously and deliberately (Zajonc, 1998; Kahneman et al., 1999; Paul Slovic et al., 2002; Epstein, 2003).

The preceding discussion establishes a dimension of accessibility. At one end of this dimension we find operations that have the characteristics of perception and of the intuitive System 1: they are rapid, automatic, and effortless. At the other end are slow, serial, and effortful operations that people need a special reason to undertake. Accessibility is a continuum, not a dichotomy, and some effortful operations demand more effort than others. Some of the determinants of accessibility are probably genetic; others develop through experience. The acquisition of skill gradually increases the accessibility of useful responses and of productive ways to organize information, until skilled performance becomes almost effortless. This effect of practice is not limited to motor skills. A master chess player does not see the same board as the novice, and visualizing the tower in an array of blocks would also become virtually effortless with prolonged practice.

The impressions that become accessible in any particular situation are mainly determined, of course, by the actual properties of the object of judgment: it is easier to see a tower in Figure 2a than in Figure 2b, because the tower in the latter is only virtual. Physical salience also determines accessibility: if a large green letter and a small blue letter are shown at the same time, "green" will come to mind first. However, salience can be overcome by deliberate attention: an instruction to look for the small object will enhance the accessibility of all its features.

Analogous effects of salience and of spontaneous and voluntary attention occur with more abstract stimuli. For example, the statements "Team A beat team B" and "Team B lost to



FIGURE 4. AN EFFECT OF CONTEXT ON ACCESSIBILITY

team A" convey the same information, but because each sentence draws attention to its grammatical subject, they make different thoughts accessible. Accessibility also reflects temporary states of associative activation. For example, the mention of a familiar social category temporarily increases the accessibility of the traits associated with the category stereotype, as indicated by a lowered threshold for recognizing behaviors as indications of these traits (Susan T. Fiske, 1998).

As designers of billboards know well, motivationally relevant and emotionally arousing stimuli spontaneously attract attention. Billboards are useful to advertisers because paying attention to an object makes all its features accessible-including those that are not linked to its primary motivational or emotional significance. The "hot" states of high emotional and motivational arousal greatly increase the accessibility of thoughts that relate to the immediate emotion and to the current needs, and reduce the accessibility of other thoughts (George Loewenstein, 1996, 2000; Jon Elster, 1998). An effect of emotional significance on accessibility was demonstrated in an important study by Yuval Rottenstreich and Christopher K. Hsee (2001), which showed that people are less sensitive to variations of probability when valuing chances to receive emotionally loaded outcomes (kisses and electric shocks) than when the outcomes are monetary.

Figure 4 (adapted from Jerome S. Bruner and A. Leigh Minturn, 1955) includes a standard demonstration of the effect of context on accessibility. An ambiguous stimulus that is perceived as a letter within a context of letters is instead seen as a number when placed within a context of numbers. More generally, expectations (conscious or not) are a powerful determinant of accessibility.

Another important point that Figure 4 illustrates is the complete suppression of ambiguity in conscious perception. This aspect of the demonstration is spoiled for the reader who sees the two versions in close proximity, but when the two lines are shown separately, observers will not spontaneously become aware of the alternative interpretation. They "see" the interpretation of the object that is the most likely in its context, but have no subjective indication that it could be seen differently. Ambiguity and uncertainty are suppressed in intuitive judgment as well as in perception. Doubt is a phenomenon of System 2, an awareness of one's ability to think incompatible thoughts about the same thing. The central finding in studies of intuitive decisions, as described by Klein (1998), is that experienced decision makers working under pressure (e.g., firefighting company captains) rarely need to choose between options because, in most cases, only a single option comes to mind.

The compound cognitive system that has been sketched here is an impressive computational device. It is well-adapted to its environment and has two ways of adjusting to changes: a short-term process that is flexible and effortful, and a long-term process of skill acquisition that eventually produces highly effective responses at low cost. The system tends to see what it expects to see-a form of Bayesian adaptation—and it is also capable of responding effectively to surprises. However, this marvelous creation differs in important respects from another paragon, the rational agent assumed in economic theory. Some of these differences are explored in the following sections, which review several familiar results as effects of accessibility. Possible implications for theorizing in behavioral economics are explored along the way.

III. Changes or States: Prospect Theory

A general property of perceptual systems is that they are designed to enhance the accessibility of changes and differences. Perception is *reference-dependent:* the perceived attributes of a focal stimulus reflect the contrast between that stimulus and a context of prior and concurrent stimuli. This section will show that



FIGURE 5. REFERENCE-DEPENDENCE IN THE PERCEPTION OF BRIGHTNESS

intuitive evaluations of outcomes are also reference-dependent.

The role of prior stimulation is familiar in the domain of temperature. Immersing the hand in water at 20°C will feel pleasantly warm after prolonged immersion in much colder water, and pleasantly cool after immersion in much warmer water. Figure 5 illustrates referencedependence in vision. The two enclosed squares have the same luminance, but they do not appear equally bright. The point of the demonstration is that the brightness of an area is not a single-parameter function of the light energy that reaches the eye from that area, just as the experience of temperature is not a single-parameter function of the temperature to which one is currently exposed. An account of perceived brightness or temperature also requires a parameter for a reference value (often called adaptation level), which is influenced by the context of current and prior stimulation.

From the vantage point of a student of perception, it is quite surprising that in standard economic analyses the utility of decision outcomes is assumed to be determined entirely by the final state of endowment, and is therefore reference-independent. In the context of risky choice, this assumption can be traced to the brilliant essay that first defined a theory of expected utility (Daniel Bernoulli, 1738). Bernoulli assumed that states of wealth have a specified utility, and proposed that the decision rule for choice under risk is to maximize the expected utility of wealth (the moral expectation). The language of Bernoulli's essay is prescriptive—it speaks of what is sensible or reasonable to do—but the theory was also intended as a description of the choices of reasonable men (Gerd Gigerenzer et al., 1989). As in most modern treatments of decision-making, Bernoulli's essay does not acknowledge any tension between prescription and description. The proposition that decision makers evaluate outcomes by the utility of final asset positions has been retained in economic analyses for almost 300 years. This is rather remarkable, because the idea is easily shown to be wrong; I call it Bernoulli's error.

Tversky and I constructed numerous thought experiments when we began the study of risky choice that led to the formulation of prospect theory (Kahneman and Tversky, 1979). Examples such as Problems 1 and 2 below convinced us of the inadequacy of the utility function for wealth as an explanation of choice.

Problem 1 Would you accept this gamble?

> 50% chance to win \$150 50% chance to lose \$100

Would your choice change if your overall wealth were lower by \$100?

There will be few takers of the gamble in Problem 1. The experimental evidence shows that most people will reject a gamble with even chances to win and lose, unless the possible win is at least twice the size of the possible loss (e.g., Tversky and Kahneman, 1992). The answer to the second question is, of course, negative. Next consider Problem 2:

Problem 2 Which would you choose? lose \$100 with certainty or 50% chance to win \$50 50% chance to lose \$200

Would your choice change if your overall wealth were higher by \$100?

In Problem 2, the gamble appears much more attractive than the sure loss. Experimental results indicate that risk-seeking preferences are held by a large majority of respondents in problems of this kind (Kahneman and Tversky, 1979). Here again, the idea that a change of \$100 in total wealth would affect preferences cannot be taken seriously.

We examined many choice pairs of this type in our early explorations, and concluded that the very abrupt switch from risk aversion to risk seeking could not plausibly be explained by a utility function for wealth. Preferences appeared to be determined by attitudes to gains and losses, defined relative to a reference point, but Bernoulli's theory and its successors did not incorporate a reference point. We therefore proposed an alternative theory of risk, in which the carriers of utility are gains and losses-changes of wealth rather than states of wealth. One novelty of prospect theory was that it was explicitly presented as a formal descriptive theory of the choices that people actually make, not as a normative model. This was a departure from a long history of choice models that served double duty as normative logics and as idealized descriptive models.

The distinctive predictions of prospect theory follow from the shape of the value function, which is shown in Figure 6. The value function is defined on gains and losses and is



FIGURE 6. A SCHEMATIC VALUE FUNCTION FOR CHANGES

characterized by three features: (1) it is concave in the domain of gains, favoring risk aversion; (2) it is convex in the domain of losses, favoring risk seeking; (3) most important, the function is sharply kinked at the reference point, and *loss-averse*—steeper for losses than for gains by a factor of about 2–2.5 (Kahneman et al., 1991; Tversky and Kahneman, 1992).

If Bernoulli's formulation is transparently incorrect as a descriptive model of risky choices, as has been argued here, why has this model been retained for so long? The answer appears to be that the assignment of utility to wealth is an aspect of rationality, and therefore compatible with the general assumption of rationality in economic theorizing (Kahneman, 2003a). Consider Problem 3:

Two persons get their monthly report from a broker:

A is told that her wealth went from 4M to 3M

B is told that her wealth went from 1*M* to 1.1*M*

Who of the two individuals has more reason to be satisfied with her financial situation?

Who is happier today?

Problem 3

Problem 3 highlights the contrasting interpretations of utility in theories that define outcomes as states or as changes. In Bernoulli's analysis only the first of the two questions of Problem 3 is relevant, and only long-term consequences matter. Prospect theory, in contrast, is concerned with short-term outcomes, and the value function presumably reflects an anticipation of the valence and intensity of the emotions that will be experienced at moments of transition from one state to another (Kahneman, 2000a, b: Barbara Mellers, 2000). Which of these concepts of utility is more useful? The cultural norm of reasonable decision-making favors the long-term view over a concern with transient emotions. Indeed, the adoption of a broad perspective and a long-term view is an aspect of the meaning of rationality in everyday language. The finalstates interpretation of the utility of outcomes is therefore a good fit for a rational-agent model.

These considerations support the normative and prescriptive status of the Bernoullian definition of outcomes. On the other hand, an exclusive concern with the long term may be prescriptively sterile, because the long term is not where life is lived. Utility cannot be divorced from emotion, and emotions are triggered by changes. A theory of choice that completely ignores feelings such as the pain of losses and the regret of mistakes is not only descriptively unrealistic, it also leads to prescriptions that do not maximize the utility of outcomes as they are actually experienced that is, utility as Bentham conceived it (Kahneman, 1994, 2000a; Kahneman et al., 1997).

Bernoulli's error—the idea that the carriers of utility are final states—is not restricted to decision-making under risk. Indeed, the incorrect assumption that initial endowments do not matter is the basis of Coase's theorem and of its multiple applications (Kahneman et al., 1990). The error of reference-independence is built into the standard representation of indifference maps. It is puzzling to a psychologist that these maps do not include a representation of the decision maker's current holdings of various goods—the counterpart of the reference point in prospect theory. The parameter is not included, of course, because consumer theory assumes that it does not matter.

The core idea of prospect theory—that the value function is kinked at the reference point and loss averse—became useful to economics

when Thaler (1980) used it to explain riskless choices. In particular, loss aversion explained a violation of consumer theory that Thaler identified and labeled the "endowment effect": the selling price for consumption goods is much higher than the buying price, often by a factor of 2 or more. The value of a good to an individual appears to be higher when the good is viewed as something that could be lost or given up than when the same good is evaluated as a potential gain (Kahneman et al., 1990, 1991; Tversky and Kahneman, 1991).

When half the participants in an experimental market were randomly chosen to be endowed with a good (a mug) and trade was allowed, the volume of trade was about half the amount that would be predicted by assuming that value was independent of initial endowment (Kahneman et al., 1990). Transaction costs did not explain this counterexample to the Coase theorem, because the same institution produced no indication of reluctance to trade when the objects of trade were money tokens. The results suggest that the participants in these experiments did not value the mug as an object they could have and consume, but as something they could get, or give up. Interestingly, John A. List (2003a, b) found that the magnitude of the endowment effect was substantially reduced for participants with intense experience in the trading of sportscards. Experienced traders (who are also consumers) showed less reluctance to trade one good for another-not only sportscards, but also mugs and other goods—as if they had learned to base their choice on long-term value, rather than on the immediate emotions associated with getting or giving up objects.

Reference-dependence and loss aversion help account for several phenomena of choice. The familiar observation that out-of-pocket losses are valued much more than opportunity costs is readily explained, if these outcomes are evaluated on different limbs of the value function. The distinction between "actual" losses and losses of opportunities is recognized in many ways in the law (David Cohen and Jack L. Knetsch, 1992) and in lay intuitions about rules of fairness in the market (Kahneman et al., 1986). Loss aversion also contributes to the well-documented status-quo bias (William Samuelson and Richard Zeckhauser, 1988). Because the reference point is usually the status quo, the properties of alternative options are evaluated as advantages or disadvantages

DECEMBER 2003

relative to the current situation, and the disadvantages of the alternatives loom larger than their advantages. Other applications of the concept of loss aversion are documented in several chapters in Kahneman and Tversky (2000).

IV. Framing Effects

In the display of blocks in Figure 2, the same property (the total height of a set of blocks) was highly accessible in one display and not in another, although both displays contained the same information. This observation is entirely unremarkable—it does not seem shocking that some attributes of a stimulus are automatically perceived while others must be computed, or that the same attribute is perceived in one display of an object but must be computed in another. In the context of decision-making, however, similar observations raise a significant challenge to the rational-agent model.

The assumption that preferences are not affected by inconsequential variations in the description of outcomes has been called extensionality (Kenneth J. Arrow, 1982) and invariance (Tversky and Kahneman, 1986), and is considered an essential aspect of rationality. Invariance is violated in *framing effects*, where extensionally equivalent descriptions lead to different choices by altering the relative salience of different aspects of the problem. Tversky and Kahneman (1981) introduced their discussion of framing effects with the following problem:

The Asian disease

Imagine that the United States is preparing for the outbreak of an unusual Asian disease, which is expected to kill 600 people. Two alternative programs to combat the disease have been proposed. Assume that the exact scientific estimates of the consequences of the programs are as follows:

If Program A is adopted, 200 people will be saved

If Program B is adopted, there is a one-third probability that 600 people will be saved and a two-thirds probability that no people will be saved In this version of the problem, a substantial majority of respondents favor Program A, indicating risk aversion. Other respondents, selected at random, receive a question in which the same cover story is followed by a different description of the options:

If Program A' is adopted, 400 people will die

If Program B' is adopted, there is a onethird probability that nobody will die and a two-thirds probability that 600 people will die

A substantial majority of respondents now favor Program B', the risk-seeking option. Although there is no substantive difference between the versions, they evoke different associations and evaluations. This is easiest to see in the certain option, because outcomes that are certain are overweighted relative to outcomes of high or intermediate probability (Kahneman and Tversky, 1979). Thus, the certainty of saving people is disproportionately attractive, while accepting the certain death of people is disproportionately aversive. These immediate affective responses respectively favor A over B and B' over A'. As in Figures 2a and 2b, the different representations of the outcomes highlight some features of the situation and mask others.

In an essay about the ethics of policy, Thomas C. Schelling (1984) presented a compellingly realistic example of the dilemmas raised by framing. Schelling reports asking his students to evaluate a tax policy that would allow a larger child exemption to the rich than to the poor. Not surprisingly, his students found this proposal outrageous. Schelling then pointed out that the default case in the standard tax table is a childless family, with special adjustments for families with children, and led his class to agree that the existing tax schedule could be rewritten with a family with two children as the default case. In this formulation, childless families would pay a surcharge. Should this surcharge be as large for the poor as for the rich? Of course not. The two versions of the question about how to treat the rich and the poor both trigger an intuitive preference for protecting the poor, but these preferences are incoherent. Schelling's problem highlights an important point. Framing effects are not a laboratory curiosity, but a ubiquitous reality. The tax table must be framed one way or another, and each frame will increase the accessibility of some responses and make other responses less likely.

There has been considerable interest among behavioral economists in a particular type of framing effect, where a choice between two options A and B is affected by designating either A or B as a default option. The option designated as the default has a large advantage in such choices, even for decisions that have considerable significance. Eric J. Johnson et al. (1993) described a compelling example. The states of Pennsylvania and New Jersey both offer drivers a choice between an insurance policy that allows an unconstrained right to sue, and a less expensive policy that restricts the right to sue. The unconstrained right to sue is the default in Pennsylvania, the opposite is the default in New Jersey, and the takeup of full coverage is 79 percent and 30 percent in the two states, respectively. Johnson and Daniel G. Goldstein (2003) estimate that Pennsylvania drivers spend 450 million dollars annually on full coverage that they would not purchase if their choice were framed as it is for New Jersey drivers.

Johnson and Goldstein (2003) also compared the proportions of the population enrolled in organ donation programs in seven European countries in which enrollment was the default and four in which nonenrollment was the default. Averaging over countries, enrollment in donor programs was 97.4 percent when this was the default option, 18 percent otherwise. The passive acceptance of the formulation given has significant consequences in this case, as it does in other recent studies where the selection of the default on the form that workers completed to set their 401(k) contributions dominated their ultimate choice (Brigitte Madrian and Dennis Shea, 2001; James J. Choi et al., 2002).

The basic principle of framing is the passive acceptance of the formulation given. Because of this passivity, people fail to construct a canonical representation for all extensionally equivalent descriptions of a state of affairs. They do not spontaneously compute the height of a tower that could be built from an array of blocks, and they do not spontaneously transform the representation of puzzles or decision problems. Obviously, no one is able to recognize " 137×24 " and "3,288" as "the same" number without going through some elaborate computations. Invariance cannot be achieved by a finite mind.

The impossibility of invariance raises significant doubts about the descriptive realism of rational-choice models (Tversky and Kahneman, 1986). Absent a system that reliably generates appropriate canonical representations, intuitive decisions will be shaped by the factors that determine the accessibility of different features of the situation. Highly accessible features will influence decisions, while features of low accessibility will be largely ignored—and the correlation between accessibility and reflective judgments of relevance in a state of complete information is not necessarily high.

A particularly unrealistic assumption of the rational-agent model is that agents make their choices in a comprehensively inclusive context, which incorporates all the relevant details of the present situation, as well as expectations about all future opportunities and risks. Much evidence supports the contrasting claim that people's views of decisions and outcomes are normally characterized by "narrow framing" (Kahneman and Daniel Lovallo, 1993), and by the related notions of "mental accounting" (Thaler, 1985, 1999) and "decision bracketing" (Daniel Read et al., 1999).

The following are some examples of the prevalence of narrow framing. The decision of whether or not to accept a gamble is normally considered as a response to a single opportunity, not as an occasion to apply a general policy (Gideon Keren and Willem A. Wagenaar, 1987; Tversky and Donald A. Redelmeier, 1992; Kahneman and Lovallo, 1993; Shlomo Benartzi and Thaler, 1999). Investors' decisions about particular investments appear to be considered in isolation from the remainder of the investor's portfolio (Nicholas Barberis et al., 2003). The time horizon that investors adopt for evaluating their investments appears to be unreasonably short-an observation that helps explain the equity-premium puzzle (Benartzi and Thaler, 1995). Finally, the prevalence of the gain/loss framing of outcomes over the wealth frame, which was discussed in the previous section, can now be seen as an instance of narrow

framing. A shared feature of all these examples is that decisions made in narrow frames depart far more from risk neutrality than decisions that are made in a more inclusive context.

The prevalence of narrow frames is an effect of accessibility, which can be understood by referring to the displays of blocks in Figure 2. The same set of blocks is framed as a tower in Figure 2a, and as a flat array in Figure 2b. Although it is possible to "see" a tower in Figure 2b, it is much easier to do so in Figure 2a. Narrow frames generally reflect the structure of the environment in which decisions are made. The choices that people face arise one at a time, and the principle of passive acceptance suggests that they will be considered as they arise. The problem at hand and the immediate consequences of the choice will be far more accessible than all other considerations, and as a result decision problems will be framed far more narrowly than the rational model assumes.

V. Attribute Substitution: A Model of Judgment Heuristics

The first research program that Tversky and I undertook together consisted of a series of studies of various types of judgment about uncertain events, including numerical predictions and assessments of the probabilities of hypotheses. Our conclusion in a review of this work was that "people rely on a limited number of heuristic principles which reduce the complex tasks of assessing probabilities and predicting values to simpler judgmental operations. In general, these heuristics are quite useful, but sometimes they lead to severe and systematic errors" (Tversky and Kahneman, 1974, p. 1124). The article introduced three heuristics-representativeness, availability, and anchoring-that were used to explain a dozen systematic biases in judgment under uncertainty, including nonregressive prediction, neglect of base-rate information, overconfidence, and overestimates of the frequency of events that are easy to recall. Some of the biases were identified by systematic errors in estimates of known quantities and statistical facts. Other biases were defined by discrepancies between the regularities of intuitive judgments and the principles of probability theory, Bayesian inference, and regression analysis.



FIGURE 7. AN ILLUSION OF ATTRIBUTE SUBSTITUTION Source: Photo by Lenore Shoham, 2003.

Kahneman and Frederick (2002) recently revisited the early studies of judgment heuristics, and proposed a formulation in which the reduction of complex tasks to simpler operations is achieved by an operation of *attribute substitution*. "Judgment is said to be mediated by a heuristic when the individual assesses a specified *target attribute* of a judgment object by substituting another property of that object—the *heuristic attribute*—which comes more readily to mind" (p. 53). Unlike the early work, Kahneman and Frederick's conception of heuristics is not restricted to the domain of judgment under uncertainty.

For a perceptual example of attribute substitution, consider the question: "What are the sizes of the two horses in Figure 7, as they are drawn on the page?" The images are in fact identical in size, but the figure produces a compelling illusion. The target attribute that observers intend to evaluate is objective twodimensional size, but they are unable to do this veridically. Their judgments map an impression of three-dimensional size (the heuristic attribute) onto units of length that are appropriate to the target attribute, and scaled to the size of the page. This illusion is caused by the differential accessibility of competing interpretations of the image. An impression of threedimensional size is the only impression of size that comes to mind for naïve observers—painters and experienced photographers are able to do better—and it produces an illusion in the perception of picture size.

A study by Fritz Strack et al. (1988) illustrates the role of attribute substitution in a different context. College students responded to a survey which included the two following questions in immediate succession: "How happy are you with your life in general?" and "How many dates did you have last month?" The correlation between the two questions was 0.12 when they appeared in the order shown. Among respondents who received the same questions in reverse order, the correlation was 0.66. The psychological interpretation of the high correlation¹ is inferential, but straightforward. The dating question undoubtedly evoked in many respondents an emotionally charged evaluation of their romantic life. This evaluation was highly accessible when the question about happiness was encountered next, and it was mapped onto the scale of general happiness. In the interpretation offered here, the respondents answered the happiness question by reporting what came to their mind, and failed to notice that they were answering a question that had not been asked—a cognitive illusion that is analogous to the visual illusion of Figure 7.

The most direct evidence for attribute substitution was reported by Kahneman and Tversky (1973), in a task of categorical prediction. There were three experimental groups in their experiment. Participants in a *base-rate* group evaluated the relative frequencies of graduate students in nine categories of specialization.² Mean estimates ranged from 20 percent for Humanities and Education to 3 percent for Library Science.

Two other groups of participants were shown the same list of areas of graduate specialization, and the following description of a fictitious graduate student. Tom W. is of high intelligence, although lacking in true creativity. He has a need for order and clarity, and for neat and tidy systems in which every detail finds its appropriate place. His writing is rather dull and mechanical, occasionally enlivened by somewhat corny puns and by flashes of imagination of the sci-fi type. He has a strong drive for competence. He seems to have little feel and little sympathy for other people and does not enjoy interacting with others. Self-centered, he nonetheless has a deep moral sense.

Participants in a *similarity* group ranked the nine fields by the degree to which Tom W. "resembles a typical graduate student" (in that field). The description of Tom W. was deliberately constructed to make him more representative of the less populated fields, and this manipulation was successful: the correlation between the averages of representativeness rankings and of estimated base rates was -0.62. Participants in the *probability* group ranked the nine fields according to the likelihood that Tom W. would have specialized in each. The respondents in the latter group were graduate students in psychology at major universities. They were told that the personality sketch had been written by a psychologist when Tom W. was in high school, on the basis of personality tests of dubious validity. This information was intended to discredit the description as a source of valid information.

The statistical logic is straightforward. A description based on unreliable information must be given little weight, and predictions made in the absence of valid evidence must revert to base rates. This reasoning implies that judgments of probability should be highly correlated with the corresponding base rates in the Tom W. problem.

The psychology of the task is also straightforward. The similarity of Tom W. to various stereotypes is a highly accessible natural assessment, whereas judgments of probability are difficult. The respondents are therefore expected to substitute a judgment of similarity (representativeness) for the required judgment of probability. The two instructions—to rate similarity or

¹ The observed value of 0.66 underestimates the true correlation between the variables of interest, because of measurement error in all variables.

² The categories were Business Administration; Computer Science; Engineering; Humanities and Education; Law; Library Science; Medicine; Physical and Life Sciences; Social Sciences and Social Work.



FIGURE 8. TWO TESTS OF ATTRIBUTE SUBSTITUTION IN A PREDICTION TASK

probability—should therefore elicit similar judgments.

The scatterplot of the mean judgments of the two groups is presented in Figure 8a. As the figure shows, the correlation between judgments of probability and similarity is nearly perfect (0.98). The correlation between judgments of probability and base rates is -0.63. The results are in perfect accord with the hypothesis of attribute substitution. They also confirm a bias of *base-rate neglect* in this prediction task. The results are especially compelling because the responses were rankings. The large variability of the average rankings of both attributes indicates highly consensual responses, and nearly total overlap in the systematic variance.

Figure 8b shows the results of another study in the same design, in which respondents were shown the description of a woman named Linda, and a list of eight possible outcomes describing her present employment and activities. The two critical items in the list were #6 ("Linda is a bank teller") and the conjunction item #8 ("Linda is a bank teller and active in the feminist movement"). The other six possibilities were unrelated and miscellaneous (e.g., elementary school teacher, psychiatric social worker). As in the Tom W. problem, some respondents ranked the eight outcomes by the similarity of Linda to the category prototypes; others ranked the same outcomes by probability.

Linda is 31 years old, single, outspoken and very bright. She majored in philosophy. As a student she was deeply concerned with issues of discrimination and social justice and also participated in antinuclear demonstrations.

As might be expected, 85 percent of respondents in the similarity group ranked the conjunction item (#8) higher than its constituent, indicating that Linda resembles the image of a feminist bank teller more than she resembles a stereotypical bank teller. This ordering of the two items is quite reasonable for judgments of similarity. However, it is much more problematic that 89 percent of respondents in the probability group also ranked the conjunction higher than its constituent. This pattern of probability judgments violates monotonicity, and has been called the "conjunction fallacy" (Tversky and Kahneman, 1983).

The observation that biases of judgment are systematic was quickly recognized as relevant to the debate about the assumption of rationality

1463

in economics (see, e.g., Peter A. Diamond, 1977; David M. Grether, 1978; Howard Kunreuther, 1979; Arrow, 1982). There has also been some discussion of the role of specific judgment biases in economic phenomena, especially in finance (e.g., Werner F. M. De Bondt and Thaler, 1985; Robert J. Shiller, 2000; Andrei Shleifer, 2000; Matthew Rabin, 2002). Recent extensions of the notion of heuristics to the domain of affect may be of particular relevance to the conversation between psychology and economics, because they bear on the core concept of a preference. As was noted earlier, affective valence is a natural assessment, which is automatically computed and always accessible. This basic evaluative attribute (good/bad, like/ dislike, approach/avoid) is therefore a candidate for substitution in any task that calls for a favorable or unfavorable response. Slovic and his colleagues (see, e.g., Slovic et al., 2002) introduced the concept of an affect heuristic. They showed that affect (liking or disliking) is the heuristic attribute for numerous target attributes, including the evaluation of the costs and benefits of various technologies, the safe concentration of chemicals, and even the predicted economic performance of various industries. In an article aptly titled "Risk as Feelings," Loewenstein et al. (2001) documented the related proposition that beliefs about risk are often expressions of emotion.

If different target attributes are strongly influenced by the same affective reaction, the dimensionality of decisions and judgments about valued objects may be expected to be unreasonably low. Indeed, Melissa L. Finucane et al. (2000) found that people's judgments of the costs and benefits of various technologies are negatively correlated, especially when the judgments are made under time pressure. A technology that is liked is judged to have low costs and large benefits. These judgments are surely biased, because the correlation between costs and benefits is generally positive in the world of real choices. In the same vein, Kahneman et al. (1997) presented evidence that different responses to public goods (e.g., willingness to pay, ratings of moral satisfaction for contributing) yielded essentially interchangeable rankings of a set of policy issues. Here again, a basic affective response appeared to be the common factor.

Kahneman et al. (1997) suggested that peo-

ple's decisions often express affective evaluations (attitudes), which do not conform to the logic of economic preferences. To understand preferences, then, we may need to understand the psychology of emotions. And we cannot take it for granted that preferences that are controlled by the emotion of the moment will be internally coherent, or even reasonable by the cooler criteria of reflective reasoning. In other words, the preferences of System 1 are not necessarily consistent with the preferences of System 2. The next section will show that some choices are not appropriately sensitive to variations of quantity and cost-and are better described as expressions of an affective response than as economic preferences.

VI. Prototype Heuristics

The results summarized in Figure 8 showed that the judgments that subjects made about the Tom W. and Linda problems substituted the more accessible attribute of similarity (representativeness) for the required target attribute of probability. The goal of the present section is to embed the representativeness heuristic in a broader class of *prototype heuristics*, which share a common psychological mechanism—the representation of categories by their prototypes—and a remarkably consistent pattern of biases.

In the display of lines in Figure 3, the average (typical) length of the lines was highly accessible, but the sum of their lengths was not. Both observations are quite general. Classic psychological experiments have established the following proposition: whenever we look at or think about a set (ensemble, category) which is sufficiently homogeneous to have a prototype, information about the prototype is automatically accessible (Michael I. Posner and Stephen W. Keele, 1968; Eleanor Rosch and Carolyn B. Mervis, 1975). The prototype of a set is characterized by the average values of the salient properties of its members. The high accessibility of prototype information serves an important adaptive function. It allows new stimuli to be categorized efficiently, by comparing their features to those of category prototypes.³ For

³ Stored information about individual exemplars also contributes to categorization.

example, the stored prototype of a set of lines allows a quick decision about a new line—does it belong with the set? There is no equally obvious function for the automatic computation of sums.

The low accessibility of sums and the high accessibility of prototypes have significant consequences in tasks that involve judgments of sets, as in the following examples:

- (i) category prediction (e.g., the probability that the category of bank tellers contains Linda as a member);
- (ii) pricing a quantity of public or private goods (e.g., the personal dollar value of saving a certain number of migratory birds from drowning in oil ponds);
- (iii) global evaluation of a past experience that extended over time (e.g., *the overall aversiveness of a painful medical procedure*);
- (iv) assessment of the support that a sample of observations provides for a hypothesis (e.g., the probability that a sample of colored balls has been drawn from one specified urn rather than another).

The objects of judgment in these tasks are sets or categories, and the target attributes have a common logical structure. Extensional attributes are governed by a general principle of conditional adding, which dictates that each element within the set adds to the overall value an amount that depends on the elements already included. In simple cases, the value is additive: the total length of the set of lines in Figure 3 is just the sum of their separate lengths. In other cases, each positive element of the set increases the aggregate value, but the combination rule is nonadditive (typically subadditive).⁴ The attributes of the category prototype are not extensional-they are averages, whereas extensional attributes are akin to sums.

The preceding argument leads to the hypothesis that tasks that require the assessment of extensional variables will be relatively difficult, and that intuitive responses may be generated by substituting an attribute of the prototype for the extensional target attribute. Prototype heuristics involve a target attribute that is extensional, and a heuristic attribute which is a characteristic of the category prototype. Prototype heuristics are associated with two major biases, which generalize the biases of representativeness that were introduced in the preceding section:

- (i) Violations of monotonicity. Adding elements to a set may lower the average and cause the judgment of the target variable to decrease, contrary to the logic of extensional variables. The prevalent judgment that Linda is less likely to be a bank teller than to be a feminist bank teller illustrates this bias.
- (ii) Extension neglect. Other things equal, an increase in the extension of a category will increase the value of its extensional attributes, but leave unchanged the values of its prototype attributes. The apparent neglect of the base rates of areas of specialization in judgments about Tom W. is an example.

Studies that have examined the two biases in different contexts are described next.

A. Pricing Goods

The price of a set of goods is an extensional variable. If price is evaluated by the attractiveness of a prototypical element of the set, violations of monotonicity and extension neglect are predicted.

Scope Neglect.—Complete or almost complete neglect of extension has often been observed in studies of the willingness to pay for public goods, where the effect is called "neglect of scope." The best known example is a study by William H. Desvousges et al. (1993) in which respondents indicated their willingness to contribute money to prevent the drowning of migratory birds. The number of birds that would be saved was varied for different subsamples. The estimated amounts that households were willing to pay were \$80, \$78, and \$88, to save 2,000, 20,000, or 200,000 birds, respectively.

⁴ If the judgment is monotonically related to an additive scale (such as the underlying count of the number of birds), the formal structure is known in the measurement literature as an "extensive structure" (R. Duncan Luce et al., 1990, Ch. 3). There also may be attributes that lack an underlying additive scale, in which case the structure is known in the literature as a "positive concatenation structure" (Luce et al., 1990, Ch. 19, volume 3, p. 38).

The target attribute in this case is willingness to pay (WTP), and the heuristic attribute appears to be the emotion associated with the image of a bird drowning in oil, or perhaps with the image of a bird being saved from drowning (Kahneman et al., 1999).

Frederick and Baruch Fischhoff (1998) reviewed numerous demonstrations of such scope neglect in studies of willingness to pay for public goods. For example, Kahneman and Knetsch found that survey respondents in Toronto were willing to pay almost as much to clean up the lakes in a small region of Ontario or to clean up all the lakes in that province (reported by Kahneman, 1986). The issue of scope neglect is central to the application of the contingent valuation method (CVM) in the assessment of the economic value of public goods, and it has been hotly debated (see, e.g., Richard T. Carson, 1997). The proponents of CVM have reported experiments in which there was some sensitivity to scope, but even these effects are minute, far too small to satisfy the economic logic of pricing (Diamond, 1996; Kahneman et al., 1999).

Violations of Monotonicity.-List (2002) reported an experiment that confirmed, in a real market setting, violations of dominance that Hsee (1998) had previously reported in a hypothetical pricing task. In List's experiment, traders of sportscards assigned significantly higher value to a set of ten sportscards labeled "Mint/ near mint condition" than to a set that included the same ten cards and three additional cards described as "poor condition." In a series of follow-up experiments, Jonathan E. Alevy et al. (2003) also confirmed an important difference (originally suggested by Hsee) between the prices that people will pay when they see only one of the goods (separate evaluation), or when they price both goods at the same time (joint evaluation). The goods were similar to those used in List's experiment. The predicted violation of dominance was observed in separate evaluation, especially for relatively inexperienced market participants. These individuals bid an average of \$4.05 for the small set of cards, and only \$1.82 for the larger set. The violations of dominance were completely eliminated in the joint evaluation condition, where the bids for the small and large sets averaged \$2.89 and \$3.32, respectively. Alevy et al. (2003) noted that System 1 appears to dominate responses in separate evaluation, whereas System 2 conforms to the dominance rule when given a chance to do so. There was a definite effect of market experience, both in this study and in List (2002): the bids of highly experienced traders also showed violations of monotonicity in separate evaluation, but the effect was much smaller.

B. Evaluations of Extended Episodes

The global utility of an experience that extends over time is an extensional attribute (Kahneman, 1994, 2000a, b; Kahneman et al., 1997), and the duration of the experience is a measure of its extension. The corresponding prototype attribute is the experienced utility associated with a representative moment of the episode. As predicted by attribute substitution, global evaluations of the episode exhibit both duration neglect and violations of monotonicity.

Duration Neglect.-In a study described by Redelmeier and Kahneman (1996), patients undergoing colonoscopy reported the intensity of pain every 60 seconds during the procedure (see Figure 9), and subsequently provided a global evaluation of the pain they had suffered. The correlation of global evaluations with the duration of the procedure (which ranged from 4 to 66 minutes in that study) was 0.03. On the other hand global evaluations were correlated (r =0.67) with an average of the pain reported at two points of time: when pain was at its peak, and just before the procedure ended. For example, patient A in Figure 9 reported a more negative evaluation of the procedure than patient B. The same pattern of *duration neglect* and Peak/ End evaluations has been observed in other studies (Barbara L. Fredrickson and Kahneman, 1993; see Kahneman, 2000a, for a discussion). These results are consistent with the hypothesis that the extended episode (which can be considered an ordered set of moments) is represented in memory by a typical moment of the experience.

Violations of Dominance.—A randomized clinical experiment was conducted following the colonoscopy study described above. For half the patients, the instrument was not immediately removed when the clinical examination



FIGURE 9. PAIN INTENSITY REPORTED BY TWO COLONOSCOPY PATIENTS

ended. Instead, the physician waited for about a minute, leaving the instrument stationary. The experience during the extra period was uncomfortable, but the procedure guaranteed that the colonoscopy never ended in severe pain. Patients reported significantly more favorable global evaluations in this experimental condition than in the control condition (Redelmeier et al., 2003).

Violations of dominance have also been confirmed in choices. Kahneman et al. (1993) exposed participants to two cold-pressor experiences, one with each hand: a "short" episode (immersion of one hand in 14°C water for 60 seconds), and a "long" episode (the short episode, plus an additional 30 seconds during which the water was gradually warmed to 15°C). When they were later asked which of the two experiences they preferred to repeat, a substantial majority chose the long trial. This pattern of choices is predicted from the Peak/End rule of evaluation that was described earlier. Similar violations of dominance were observed with unpleasant sounds of variable loudness and duration (Charles A. Schreiber and Kahneman, 2000). These violations of dominance suggest that choices between familiar experiences are made in an intuitive process of "choosing by liking." Extended episodes are represented in memory by a typical moment-and the desirability or aversiveness of the episode is dominated by the remembered utility of that moment (Kahneman, 1994). When a choice is to be made, the option that is associated with the higher remembered utility (more liked) is chosen. This mode of choice is likely to yield choices that do not maximize the utility that will actually be experienced (Kahneman et al., 1997).

C. Other Prototype Heuristics

The pattern of results observed in diverse studies of prototype heuristics suggests the need for a unified interpretation, and raises a significant challenge to treatments that deal only with one domain. A number of authors have offered competing interpretations of base-rate neglect (Leda Cosmides and John Tooby, 1996; Jonathan Jay Koehler, 1996), insensitivity to scope in WTP (Raymond Kopp, 1992), and duration neglect (Ariely and Loewenstein, 2000). In general however, these interpretations are specific to a particular task, and would not carry over to demonstrations of extension neglect in the other tasks that have been discussed. In contrast, the account offered here (and developed in greater detail by Kahneman and Frederick, 2002) is equally applicable to diverse tasks that require an assessment of an extensional target attribute.

The cases that have been discussed are only illustrations, not a comprehensive list of prototype heuristics. For example, the same form of nonextensional thinking explains why the median estimate of the annual number of murders in Detroit is twice as high as the estimate of the number of murders in Michigan (Kahneman and Frederick, 2002). It also explains why professional forecasters assigned a higher probability to "an earthquake in California causing a flood in which more than 1,000 people will drown" than to "a flood somewhere in the United States in which more than 1,000 people will drown" (Tversky and Kahneman, 1983).

As these examples illustrate, there is no guaranteed defense against violations of monotonicity. How could a forecaster who assigns a probability to a lethal flood ensure (in finite time) that there is no subset of that event which would have appeared even more probable? More generally, the results reviewed in this section suggest a profound incompatibility between the capabilities and operational rules of intuitive judgment and choice and the normative standards for beliefs and preferences. The logic of belief and choice requires accurate evaluation of extensional variables. In contrast, intuitive thinking operates with exemplars or prototypes that have the dimensionality of individual instances and lack the dimension of extension.

VII. The Boundaries of Intuitive Thinking

The judgments that people express, the actions they take, and the mistakes they commit depend on the monitoring and corrective functions of System 2, as well as on the impressions and tendencies generated by System 1. This section reviews a selection of findings and ideas about the functioning of System 2. A more detailed treatment is given in Kahneman and Frederick (2002) and Kahneman (2003b).

Judgments and choices are normally intuitive, skilled, unproblematic, and reasonably successful (Klein, 1998). The prevalence of framing effects, and other indications of superficial processing such as the bat-and-ball problem, suggest that people mostly do not think very hard and that System 2 monitors judgments quite lightly. On some occasions, however, the monitoring of System 2 will detect a potential error, and an effort will be made to correct it. The question for this section can be formulated in terms of accessibility: when do doubts about one's intuitive judgments come to mind? The answer, as usual in psychology, is a list of relevant factors.

Research has established that the ability to avoid errors of intuitive judgment is impaired by time pressure (Finucane et al., 2000), by concurrent involvement in a different cognitive task (Gilbert, 1989, 1991, 2002), by performing the task in the evening for "morning people" and in the morning for "evening people" (Galen V. Bodenhausen, 1990), and, surprisingly, by being in a good mood (Alice M. Isen et al., 1988; Herbert Bless et al., 1996). Conversely, the facility of System 2 is positively correlated with intelligence (Stanovich and West, 2002), with the trait that psychologists have labeled "need for cognition" (which is roughly whether people find thinking fun) (Eldar Shafir and Robyn A. LeBoeuf, 2002), and with exposure to statistical thinking (Richard E. Nisbett et al., 1983; Franca Agnoli and David H. Krantz, 1989; Agnoli, 1991).

The question of the precise conditions under which errors of intuition are most likely to be prevented is of methodological interest to psychologists, because some controversies in the literature on cognitive illusions are resolved when this factor is considered (see Kahneman and Frederick, 2002; Kahneman, 2003b). One of these methodological issues is also of considerable substantive interest: this is the distinction between separate evaluation and joint evaluation (Hsee, 1996).

In the separate evaluation condition of List's study of dominance violations, for example, different groups of traders bid on two sets of baseball cards; in joint evaluation each trader evaluated both sets at the same time. The results were drastically different. Violations of monotonicity, which were very pronounced in the between-groups comparison, were eliminated in the joint evaluation condition. The participants in the latter condition evidently realized that one of the sets of goods included the other, and was therefore worth more. Once they had detected the dominance relation, the participants constrained their bids to follow the rule. These decisions are mediated by System 2. Thus, there appear to be two distinct modes of choice: "choosing by liking" selects the most attractive option; "choosing by rule" conforms to an explicit constraint.

Prospect theory introduced the same distinction between modes of choice (Kahneman and Tversky, 1979). The normal process corresponds to choice by liking: the decision maker evaluates each gamble in the choice set, then selects the gamble of highest value. In prospect theory, this mode of choice can lead to the selection of a dominated option.⁵ However, the theory also introduced the possibility of choice by rule: if one option transparently dominates the other, the decision maker will select the dominant option without further evaluation. To test this model, Tversky and Kahneman (1986) constructed a pair of gambles that satisfied three criteria: (i) gamble A dominated gamble B; (ii) the prospect-theory value of B was higher than the value of A; (iii) the gambles were complex, and the dominance relation only became apparent after grouping outcomes. As expected from other framing results, most participants in the experiment evaluated the gambles as originally formulated, failed to detect the relation between them, chose the option they liked most, and exhibited the predicted violation of dominance.

The cold-pressor experiment that was described earlier (Kahneman et al., 1993) is closely analogous to the study of nontransparent dominance that Tversky and Kahneman (1986) reported. A substantial majority of participants violated dominance in a direct and seemingly transparent choice between cold-pressor experiences. However, postexperimental debriefings indicated that the dominance was not in fact transparent. The participants in the experiment did not realize that the long episode included the short one, although they did notice that the episodes differed in duration. Because they failed to detect that one option dominated the other, the majority of participants chose as people commonly do when they select an experience to be repeated: they "chose by liking," selected the option that had the higher remembered utility, and thereby agreed to expose themselves to a period of unnecessary pain (Kahneman, 1994; Kahneman et al., 1997).

The complex pattern of results in the studies of dominance in the joint-evaluation design suggests three general conclusions: (i) choices that are governed by rational rules do exist, but (ii) these choices are restricted to unusual circumstances, and (iii) the activation of the rules depends on the factors of attention and accessibility. The fact that System 2 "knows" the dominance rule and "wants" to obey it only guarantees that the rule will be followed if a potential violation is explicitly detected.

System 2 has the capability of correcting other errors, besides violations of dominance. In particular, the substitution of one attribute for another in judgment inevitably leads to errors in the weights assigned to different sources of information, and these could-at least in principle-be detected and corrected. For example, a participant in the Tom W. study (see Figure 8a) could have reasoned as follows: "Tom W. looks very much like a library science student, but there are very few of those. I should therefore adjust my impression of probability downward." Although this level of reasoning should not have been beyond the reach of the graduate students who answered the Tom W. question, the evidence shown in Figure 8 shows that few, if any, of these respondents had the idea of adjusting their predictions to allow for the different base rates of the alternative outcomes. The explanation of this result in terms of accessibility is straightforward: the experiment provided no explicit cues to the relevance of base rates.

Base-rate information was not completely ignored in experiments that provided stronger cues, though the effects of this variable were consistently too small relative to the effect of the case-specific information (Jonathan St. B. T. Evans et al., 2002). The evidence of numerous studies supports the following conclusions: (i) the likelihood that the subject will detect a misweighting of some aspect of the information depends on the salience of cues to the relevance of that factor; (ii) if the misweighting is detected, there will be an effort to correct it; (iii) the correction is likely to be insufficient, and the final judgments are therefore likely to remain anchored on the initial intuitive impression (Gretchen B. Chapman and Johnson, 2002).

Economists may be struck by the emphasis on salient cues and by the absence of financial incentives from the list of major factors that influence the quality of decisions and judgments. However, the claim that high stakes eliminate departures from rationality is not supported by a careful review of the experimental evidence (Camerer and Robin M. Hogarth, 1999). A growing literature of field research and

⁵ Cumulative prospect theory (Tversky and Kahneman, 1992) does not have this feature.

field experiments documents large and systematic mistakes in some of the most consequential financial decisions that people make, including choices of investments (Brad M. Barber and Terrance Odean, 2000; Benartzi and Thaler, 2001), and actions in the real estate market (David Genesove and Christopher J. Mayer, 2001). The daily paper provides further evidence of poor decisions with large outcomes.

The present analysis helps explain why the effects of incentives are neither large nor robust. High stakes surely increase the amount of attention and effort that people invest in their decisions. But attention and effort by themselves do not purchase rationality or guarantee good decisions. In particular, cognitive effort expended in bolstering a decision already made will not improve its quality, and the evidence suggests that the share of time and effort devoted to such bolstering may increase when the stakes are high (Jennifer S. Lerner and Philip E. Tetlock, 1999). Effort and concentration are likely to bring to mind a more complete set of considerations, but the expansion may yield an inferior decision unless the weighting of the secondary considerations is appropriately low. In some instances—including tasks that require predictions of one's future tastes-too much cognitive effort actually lowers the quality of performance (Wilson and Jonathan W. Schooler, 1991). Klein (2003, Ch. 4) has argued that there are other situations in which skilled decision makers do better when they trust their intuitions than when they engage in detailed analysis.

VIII. Concluding Remarks

The rational agent of economic theory would be described, in the language of the present treatment, as endowed with a single cognitive system that has the logical ability of a flawless System 2 and the low computing costs of System 1. Theories in behavioral economics have generally retained the basic architecture of the rational model, adding assumptions about cognitive limitations designed to account for specific anomalies. For example, the agent may be rational except for discounting hyperbolically, evaluating outcomes as changes, or a tendency to jump to conclusions.

The model of the agent that has been presented here has a different architecture, which may be more difficult to translate into the theoretical language of economics. The core ideas of the present treatment are the two-system structure, the large role of System 1 and the extreme context-dependence that is implied by the concept of accessibility. The central characteristic of agents is not that they reason poorly but that they often act intuitively. And the behavior of these agents is not guided by what they are able to compute, but by what they happen to see at a given moment.

These propositions suggest heuristic questions that may guide attempts to predict or explain behavior in a given setting: "What would an impulsive agent be tempted to do?" "What course of action seems most natural in this situation?" The answers to these questions will often identify the judgment or course of action to which most people will be attracted. For example, it is more natural to join a group of strangers running in a particular direction than to adopt a contrarian destination. However, the two-system view also suggests that other questions should be raised: "Is the intuitively attractive judgment or course of action in conflict with a rule that the agent would endorse?" If the answer to that question is positive, then "How likely is it in the situation at hand that the relevant rule will come to mind in time to override intuition?" Of course, this mode of analysis also allows for differences between individuals. and between groups. What is natural and intuitive in a given situation is not the same for everyone: different cultural experiences favor different intuitions about the meaning of situations, and new behaviors become intuitive as skills are acquired. Even when these complexities are taken into account, the approach to the understanding and prediction of behavior that has been sketched here is simple and easy to apply, and likely to yield hypotheses that are generally plausible and often surprising. The origins of this approach are in an important intellectual tradition in psychology, which has emphasized "the power of the situation" (Lee Ross and Nisbett, 1991).

The present treatment has developed several themes: that intuition and reasoning are alternative ways to solve problems, that intuition resembles perception, that people sometimes answer a difficult question by answering an easier one instead, that the processing of information is often superficial, that categories are

represented by prototypes. All these features of the cognitive system were in our minds in some form when Amos Tversky and I began our joint work in 1969, and most of them were in Herbert Simon's mind much earlier. However, the role of emotion in judgment and decision making received less attention in that work than it had received before the beginning of the cognitive revolution in psychology in the 1950's. More recent developments have restored a central role to emotion, which is incorporated in the view of intuition that was presented here. Findings about the role of optimism in risk taking, the effects of emotion on decision weights, the role of fear in predictions of harm, and the role of liking and disliking in factual predictions—all indicate that the traditional separation between belief and preference in analyses of decision making is psychologically unrealistic.

Incorporating a common sense psychology of the intuitive agent into economic models will present difficult challenges, especially for formal theorists. It is encouraging to note, however, that the challenge of incorporating the first wave of psychological findings into economics appeared even more daunting 20 years ago, and that challenge has been met with considerable success.

REFERENCES

- Agnoli, Franca. "Development of Judgmental Heuristics and Logical Reasoning: Training Counteracts the Representativeness Heuristic." *Cognitive Development*, April–June 1991, 6(2), pp. 195–217.
- Agnoli, Franca and Krantz, David H. "Suppressing Natural Heuristics by Formal Instruction: The Case of the Conjunction Fallacy." *Cognitive Psychology*, October 1989, *21*(4), pp. 515–50.
- Alevy, Jonathan E.; List, John A. and Adamowicz, Wiktor. "More is Less: Preference Reversals and Non-Market Valuations." Working paper, University of Maryland, 2003.
- Ariely, Dan. "Seeing Sets: Representation by Statistical Properties." *Psychological Science*, March 2001, *12*(2), pp. 157–62.
- Ariely, Dan and Loewenstein, George. "When Does Duration Matter in Judgment and Decision Making?" *Journal of Experimental Psychology: General*, December 2000, 129(4), pp. 508–23.

- Arrow, Kenneth J. "Risk Perception in Psychology and Economics." *Economic Inquiry*, January 1982, 20(1), pp. 1–9.
- **Barber, Brad M. and Odean, Terrance.** "Trading is Hazardous to Your Wealth: The Common Stock Investment Performance of Individual Investors." *Journal of Finance*, April 2000, 55(2), pp. 773–806.
- Barberis, Nicholas; Huang, Ming and Thaler, Richard H. "Individual Preferences, Monetary Gambles and the Equity Premium." National Bureau of Economic Research (Cambridge, MA) Working Paper No. W9997, May 2003.
- Bargh, John A. "The Automaticity of Everyday Life," in Robert S. Wyer, Jr., ed., *The automaticity of everyday life: Advances in social cognition*, Vol. 10. Mahwah, NJ: Erlbaum, 1997, pp. 1–61.
- Benartzi, Shlomo and Thaler, Richard H. "Myopic Loss Aversion and the Equity Premium Puzzle." *Quarterly Journal of Economics*, February 1995, *110*(1), pp. 73–92.

 - . "Naïve Diversification Strategies in Defined Contribution Saving Plans." *American Economic Review*, March 2001, *91*(1), pp. 79–98.
- Bernoulli, Daniel. "Exposition of a New Theory on the Measurement of Risk." *Econometrica*, January 1954, 22(1), pp. 23–36. (Original work published 1738.)
- Bless, Herbert; Clore, Gerald L.; Schwarz, Norbert; Golisano, Verana; Rabe, Christian and Wolk, Marcus. "Mood and the Use of Scripts: Does a Happy Mood Really Lead to Mindlessness?" Journal of Personality and Social Psychology, October 1996, 71(4), pp. 665–79.
- Bodenhausen, Galen V. "Stereotypes as Judgmental Heuristics: Evidence of Circadian Variations in Discrimination." *Psychological Science*, September 1990, 1(5), pp. 319–22.
- Bruner, Jerome S. and Minturn, A. Leigh. "Perceptual Identification and Perceptual Organization." *Journal of General Psychology*, July 1955, *53*, pp. 21–28.
- Camerer, Colin F. and Hogarth, Robin M. "The Effect of Financial Incentives." *Journal of Risk and Uncertainty*, December 1999, *19*(1–3), pp. 7–42.

- Camerer, Colin F.; Loewenstein, George and Rabin, Matthew, eds. Advances in behavioral economics. Princeton, NJ: Princeton University Press (forthcoming).
- Carson, Richard T. "Contingent Valuation Surveys and Tests of Insensitivity to Scope," in R. J. Kopp, W. W. Pommerhene, and N. Schwartz, eds., Determining the value of nonmarketed goods: Economic, psychological, and policy relevant aspects of contingent valuation methods. Boston, MA: Kluwer, 1997, pp. 127–63.
- Chaiken, Shelly and Trope, Yaacov, eds. *Dualprocess theories in social psychology*. New York: Guilford Press, 1999.
- Chapman, Gretchen B. and Johnson, Eric J. "Incorporating the Irrelevant: Anchors in Judgments of Belief and Value," in Thomas Gilovich, Dale Griffin, and Daniel Kahneman, eds., *Heuristics and biases: The psychology of intuitive thought*. New York: Cambridge University Press, 2002, pp. 120– 38.
- Choi, James J.; Laibson, David; Madrian, Brigitte and Metrick, Andrew. "Defined Contribution Pensions: Plan Rules, Participant Decisions and the Path of Least Resistance," in James M. Poterba, ed., *Tax policy and the economy*, Vol. 16. Cambridge, MA: MIT Press, 2002, pp. 67–113.
- Chong, Sang-Chul and Treisman, Anne. "Representation of Statistical Properties." Vision Research, February 2003, 43(4), pp. 393–404.
- Cohen, David and Knetsch, Jack L. "Judicial Choice and Disparities Between Measures of Economic Value." Osgoode Hall Law Review, 1992, 30(3), pp. 737–70.
- **Cosmides, Leda and Tooby, John.** "Are Humans Good Intuitive Statisticians After All? Rethinking Some Conclusions From the Literature on Judgment and Uncertainty." *Cognition*, January 1996, 58(1), pp. 1–73.
- **De Bondt, Werner F. M. and Thaler, Richard H.** "Does the Stock Market Overreact?" *Journal of Finance*, July 1985, 40(3), pp. 793–808.
- Desvousges, William H.; Johnson, F. Reed; Dunford, Richard W.; Hudson, Sara P.; Wilson, K. Nichole and Boyle, Kevin J. "Measuring Natural Resource Damages with Contingent Valuation: Tests of Validity and Reliability," in Jerry A. Hausman, ed., *Contingent valuation: A critical assessment*. Amsterdam: North-Holland, 1993, pp. 91–164.

- Diamond, Peter A. "A Framework for Social Security Analysis." *Journal of Public Economics*, December 1977, 8(3), pp. 275–98.
- . "Testing the Internal Consistency of Contingent Valuation Surveys." *Journal of Environmental Economics and Management*, May 1996, *30*(3), pp. 155–73.
- Elster, Jon. "Emotions and Economic Theory." *Journal of Economic Literature*, March 1998, 26(1), pp. 47–74.
- **Epstein, Seymour.** "Cognitive-Experiential Self-Theory of Personality," in Theodore Millon and Melvin J. Lerner, eds., *Comprehensive handbook of psychology, volume 5: Personality and social psychology.* Hoboken, NJ: Wiley & Sons, 2003, pp. 159–84.
- Evans, Jonathan St. B. T.; Handley, Simon J.; Over, David E. and Perham, Nicholas. "Background Beliefs in Bayesian Inference." *Memory and Cognition*, March 2002, *30*(2), pp. 179–90.
- Finucane, Melissa L.; Alhakami, Ali; Slovic, Paul and Johnson, Stephen M. "The Affect Heuristic in Judgments of Risks and Benefits." *Journal of Behavioral Decision Making*, January/March 2000, 13(1), pp. 1–17.
- Fiske, Susan T. "Stereotyping, Prejudice, and Discrimination," in Daniel T. Gilbert, Susan T. Fiske, and Gardner Lindzey, eds., *The handbook of social psychology*, 4th Ed., Vol. 1. New York: McGraw-Hill, 1998, pp. 357– 411.
- Frederick, Shane W. and Fischhoff, Baruch. "Scope (In)sensitivity in Elicited Valuations." *Risk, Decision, and Policy*, August 1998, 3(2), pp. 109–23.
- Fredrickson, Barbara L. and Kahneman, Daniel. "Duration Neglect in Retrospective Evaluations of Affective Episodes." *Journal of Personality and Social Psychology*, July 1993, 65(1), pp. 45–55.
- **Gawande, Atul.** Complications: A surgeon's notes on an imperfect science. New York: Metropolitan Books, 2002.
- Genesove, David and Mayer, Christopher J. "Loss Aversion and Seller Behavior: Evidence from the Housing Market." *Quarterly Journal of Economics*, November 2001, *116*(4), pp. 1233–60.
- Gigerenzer, Gerd; Swijtink, Zeno; Porter, Theodore; Daston, Lorraine; Beatty, John and Kruger, Lorenz. The empire of chance: How probability changed science and everyday

life. Cambridge: Cambridge University Press, 1989.

- Gilbert, Daniel T. "Thinking Lightly About Others: Automatic Components of the Social Inference Process," in James S. Uleman and John A. Bargh, eds., *Unintended thought*. Englewood Cliffs, NJ: Prentice-Hall, 1989, pp. 189–211.
- . "Inferential Correction," in Thomas Gilovich, Dale Griffin, and Daniel Kahneman, eds., *Heuristics and biases: The psychology of intuitive thought*. New York: Cambridge University Press, 2002, pp. 167– 84.
- Grether, David M. "Recent Psychological Studies of Behavior Under Uncertainty." *American Economic Review*, May 1978 (*Papers and Proceedings*), 68(2), pp. 70–74.
- Higgins, E. Tory. "Knowledge Activation: Accessibility, Applicability, and Salience," in E. Tory Higgins and Arie W. Kruglanski, eds., *Social psychology: Handbook of basic principles*. New York: Guilford Press, 1996, pp. 133–68.
- Hsee, Christopher K. "The Evaluability Hypothesis: An Explanation of Preference Reversals Between Joint and Separate Evaluations of Alternatives." *Organizational Behavior and Human Decision Processes*, September 1996, 67(3), pp. 247–57.
- . "Less is Better: When Low-Value Options are Valued More Highly Than High-Value Options." *Journal of Behavioral Decision Making*, June 1998, *11*(2), pp. 107–21.
- Isen, Alice M.; Nygren, Thomas E. and Ashby, F. Gregory. "Influence of Positive Affect on the Subjective Utility of Gains and Losses: It is Just Not Worth the Risk." *Journal of Personality and Social Psychology*, November 1988, 55(5), pp. 710–17.
- Johnson, Eric J. and Goldstein, Daniel G. "Do Defaults Save Lives?" Working paper, Center for Decision Sciences, Columbia University, 2003.
- Johnson, Eric J.; Hershey, John; Meszaros, Jacqueline and Kunreuther, Howard. "Framing, Probability Distortions, and Insurance Decisions." *Journal of Risk and Uncertainty*, August 1993, 7(1), pp. 35–51.

- Kahneman, Daniel. "Comment," in Ronald G. Cummings, David S. Brookshire, and William D. Schultze, eds., *Valuing environmental goods*. Totowa, NJ: Rowman and Allenheld, 1986, pp. 185–93.
- . "New Challenges to the Rationality Assumption." *Journal of Institutional and Theoretical Economics*, March 1994, *150*(1), pp. 18–36.
- . "Evaluation by Moments: Past and Future," in Daniel Kahneman and Amos Tversky, eds., *Choices, values, and frames*. New York: Cambridge University Press, 2000a, pp. 693–708.
- ."Experienced Utility and Objective Happiness: A Moment-Based Approach," in Daniel Kahneman and Amos Tversky, eds., *Choices, values, and frames.* New York: Cambridge University Press, 2000b, pp. 673–92.
- . "A Psychological Perspective on Economics." *American Economic Review*, May 2003a (*Papers and Proceedings*), 93(2), pp. 162–68.
- ______. "A Perspective on Judgment and Choice: Mapping Bounded Rationality." *American Psychologist*, September 2003b, 56(9), pp. 697–720.
- Kahneman, Daniel and Frederick, Shane. "Representativeness Revisited: Attribute Substitution in Intuitive Judgment," in Thomas Gilovich, Dale Griffin, and Daniel Kahneman, eds., *Heuristics and biases: The psychology of intuitive thought*. New York: Cambridge University Press, 2002, pp. 49–81.
- Kahneman, Daniel; Fredrickson, Barbara L.;
 Schreiber, Charles A. and Redelmeier, Donald
 A. "When More Pain is Preferred to Less: Adding a Better End." *Psychological Science*, November 1993, 4(6), pp. 401–05.
- Kahneman, Daniel; Knetsch, Jack and Thaler, Richard. "Fairness as a Constraint on Profitseeking: Entitlements in the Market." *American Economic Review*, September 1986, 76(4), pp. 728–41.
- . "Experimental Tests of the Endowment Effect and the Coase Theorem." *Journal of Political Economy*, December 1990, 98(6), pp. 1325–48.
- . "The Endowment Effect, Loss Aversion, and Status Quo Bias: Anomalies." *Journal of Economic Perspectives*, Winter 1991, 5(1), pp. 193–206.

- Kahneman, Daniel and Lovallo, Daniel. "Timid Choices and Bold Forecasts: A Cognitive Perspective on Risk Taking." *Management Science*, January 1993, *39*(1), pp. 17–31.
- Kahneman, Daniel; Ritov, Ilana and Schkade, David. "Economic Preferences or Attitude Expressions? An Analysis of Dollar Responses to Public Issues." *Journal of Risk and Uncertainty*, December 1999, *19*(1–3), pp. 203–35.
- Kahneman, Daniel; Slovic, Paul and Tversky, Amos, eds. Judgment under uncertainty: Heuristics and biases. New York: Cambridge University Press, 1982.
- Kahneman, Daniel and Tversky, Amos. "On the Psychology of Prediction." *Psychological Review*, July 1973, 80(4), pp. 237–51.
- _____, eds. *Choices, values, and frames*. New York: Cambridge University Press, 2000.
- Kahneman, Daniel; Wakker, Peter P. and Sarin, Rakesh. "Back to Bentham? Explorations of Experienced Utility." *Quarterly Journal of Economics*, May 1997, 112(2), pp. 375–405.
- Keren, Gideon and Wagenaar, Willem A. "Violations of Utility Theory in Unique and Repeated Gambles." *Journal of Experimental Psychology: Learning, Memory, and Cognition*, July 1987, *13*(3), pp. 387–91.
- Klein, Gary. Sources of power: How people make decisions. Cambridge, MA: MIT Press, 1998.

_____. Intuition at work: Why developing your gut instincts will make you better at what you do. New York: Doubleday, 2003.

- Koehler, Jonathan Jay. "The Base-Rate Fallacy Reconsidered: Descriptive, Normative, and Methodological Challenges." *Behavioral and Brain Sciences*, March 1996, *19*, pp. 1–53.
- Kopp, Raymond. "Why Existence Value Should be Used in Cost-Benefit Analysis." *Journal* of Policy Analysis and Management, Winter 1992, 11(1), pp. 123–30.
- Kunreuther, Howard. "The Changing Societal Consequences of Risks From Natural Hazards." Annals of the American Academy of Political and Social Science, May 1979, 443(443), pp. 104–16.
- Langer, Ellen J.; Blank, Arthur and Chanowitz, Benzion. "The Mindlessness of Ostensibly Thoughtful Action: The Role of 'Placebic'

Information in Interpersonal Interaction." *Journal of Personality and Social Psychology*, June 1978, *36*(6), pp. 635–42.

- LeDoux, Joseph E. "Emotion Circuits in the Brain." *Annual Review of Neuroscience*, March 2000, 23, pp. 155–84.
- Lerner, Jennifer S. and Tetlock, Philip E. "Accounting for the Effects of Accountability." *Psychological Bulletin*, March 1999, *125*(2), pp. 255–75.
- List, John A. "Preference Reversals of a Different Kind: The 'More Is Less' Phenomenon." *American Economic Review*, December 2002, 92(5), pp. 1636–43.
- ______. "Does Market Experience Eliminate Market Anomalies?" *Quarterly Journal of Economics*, February 2003a, *118*(1), pp. 47– 71.
- . "Neoclassical Theory Versus Prospect Theory: Evidence From the Marketplace." National Bureau of Economic Research (Cambridge, MA) Working Paper No. W9736, 2003b; *Econometrica*, 2004 (forthcoming).
- Loewenstein, George. "Out of Control: Visceral Influences on Behavior." Organizational Behavior and Human Decision Processes, March 1996, 65(3), pp. 272–92.
- . "Emotions in Economic Theory and Economic Behavior." *American Economic Review*, May 2000 (*Papers and Proceedings*), 90(2), pp. 426–32.
- Loewenstein, George; Weber, Elke U.; Hsee, Christopher K. and Welch, N. "Risk as Feelings." *Psychological Bulletin*, March 2001, *127*(2), pp. 267–86.
- Luce, R. Duncan; Krantz, David H.; Suppes, Patrick and Tversky, Amos. Foundations of measurement, volume 3: Representation, axiomatization, and invariance. San Diego, CA: Academic Press, 1990.
- Madrian, Brigitte and Shea, Dennis. "The Power of Suggestion: Inertia in 401(k) Participation and Savings Behavior." *Quarterly Journal of Economics*, November 2001, *116*(4), pp. 1149–87.
- Mellers, Barbara. "Choice and the Relative Pleasure of Consequences." *Psychological Bulletin*, November 2000, *126*(6), pp. 910– 24.
- Nisbett, Richard E.; Krantz, David H.; Jepson, Christopher and Kunda, Ziva. "The Use of Statistical Heuristics in Everyday Inductive

Reasoning." *Psychological Review*, October 1983, *90*(4), pp. 339–63.

- **Pashler, Harold E.** *The psychology of attention.* Cambridge, MA: MIT Press, 1998.
- **Posner, Michael I. and Keele, Stephen W.** "On the Genesis of Abstract Ideas." *Journal of Experimental Psychology*, Pt. 1, 1968, 77(3), pp. 353–63.
- **Rabin, Matthew.** "Inference by Believers in the Law of Small Numbers." *Quarterly Journal of Economics*, August 2002, *17*(3), pp. 775– 816.
- Read, Daniel; Loewenstein, George and Rabin, Matthew. "Choice Bracketing." *Journal of Risk and Uncertainty*, December 1999, *19*(1– 3), pp. 171–97.
- Redelmeier, Donald A. and Kahneman, Daniel. "Patients' Memories of Painful Medical Treatments: Real-time and Retrospective Evaluations of Two Minimally Invasive Procedures." *Pain*, July 1996, *66*(1), pp. 3–8.
- Redelmeier, Donald A.; Katz, Joel and Kahneman, Daniel. "Memories of Colonoscopy: A Randomized Trial." *Pain*, July 2003, *104*(1– 2), pp. 187–94.
- Rosch, Eleanor and Mervis, Carolyn B. "Family Resemblances: Studies in the Internal Structure of Categories." *Cognitive Psychology*, October 1975, 7(4), pp. 573–605.
- **Ross, Lee and Nisbett, Richard E.** *The person and the situation.* New York: McGraw-Hill, 1991.
- Rottenstreich, Yuval and Hsee, Christopher K. "Money, Kisses and Electric Shocks: On the Affective Psychology of Risk." *Psychological Science*, May 2001, *12*(3), pp. 185–90.
- Rozin, Paul and Nemeroff, Carol. "Sympathetic Magical Thinking: The Contagion and Similarity Heuristics," in Thomas Gilovich, Dale Griffin, and Daniel Kahneman, eds., *Heuristics and biases: The psychology of intuitive thought*. New York: Cambridge University Press, 2002, pp. 201–16.
- Samuelson, William and Zeckhauser, Richard. "Status Quo Bias in Decision Making." *Journal of Risk and Uncertainty*, March 1988, *1*(1), pp. 7–59.
- Schelling, Thomas C. Choice and consequence: Perspectives of an errant economist. Cambridge, MA: Harvard University Press, 1984.
- Schreiber, Charles A. and Kahneman, Daniel. "Determinants of the Remembered Utility of

Aversive Sounds." *Journal of Experimental Psychology: General*, March 2000, *129*(1), pp. 27–42.

- Shafir, Eldar and LeBoeuf, Robyn A. "Rationality." Annual Review of Psychology, February 2002, 53(1), pp. 419–517.
- Shiller, Robert J. Irrational exuberance. Princeton, NJ: Princeton University Press, 2000.
- Shleifer, Andrei. Inefficient markets: An introduction to behavioral finance. New York: Oxford University Press, 2000.
- Simon, Herbert A. "A Behavioral Model of Rational Choice." *Quarterly Journal of Economics*, February 1955, 69(1), pp. 99–118.
- ______. "Information Processing Models of Cognition." *Annual Review of Psychology*, February 1979, *30*, pp. 363–96.
- Simon, Herbert A. and Chase, William G. "Skill in Chess." *American Scientist*, July 1973, *61*(4), pp. 394–403.
- Sloman, Steven A. "Two Systems of Reasoning," in Thomas Gilovich, Dale Griffin, and Daniel Kahneman, eds., *Heuristics and biases: The psychology of intuitive thought*. New York: Cambridge University Press, 2002, pp. 379– 96.
- Slovic, Paul; Finucane, Melissa; Peters, Ellen and MacGregor, Donald G. "The Affect Heuristic," in Thomas Gilovich, Dale Griffin, and Daniel Kahneman, eds., *Heuristics and biases: The psychology of intuitive thought*. New York: Cambridge University Press, 2002, pp. 397–420.
- Stanovich, Keith E. and West, Richard F. "Individual Differences in Reasoning: Implications for the Rationality Debate?" *Behavioral and Brain Sciences*, October 2000, *23*(5), pp. 645–65.
 - . "Individual Differences in Reasoning: Implications for the Rationality Debate?" in Thomas Gilovich, Dale Griffin, and Daniel Kahneman, eds., *Heuristics and biases: The psychology of intuitive thought*. New York: Cambridge University Press, 2002, pp. 421– 40.
- Strack, Fritz; Martin, Leonard and Schwarz, Norbert. "Priming and Communication: Social Determinants of Information Use in Judgments of Life Satisfaction." European Journal of Social Psychology, October– November 1988, 18(5), pp. 429–42.
- Thaler, Richard H. "Toward a Positive Theory of Consumer Choice." *Journal of Economic*

Behavior and Organization, March 1980, *1*(1), pp. 36–90.

. "Mental Accounting and Consumer Choice." *Marketing Science*, Summer 1985, 4(3), pp. 199–214.

_____. *Quasi rational economics*. New York: Russell Sage Foundation, 1991.

- _____. *The winner's curse: Paradoxes and anomalies of economic life*. New York: Free Press, 1992.
- . "Mental Accounting Matters." *Journal* of Behavioral Decision Making, July 1999, *12*(3), pp. 183–206.
- Tversky, Amos and Kahneman, Daniel. "Judgment under Uncertainty: Heuristics and Biases." *Science*, September 1974, *185*(4157), pp. 1124–31.
- ______. "The Framing of Decisions and the Psychology of Choice." *Science*, January 1981, *211*(4481), pp. 453–58.
- . "Extensional Versus Intuitive Reasoning: The Conjunction Fallacy in Probability Judgment." *Psychological Review*, October 1983, *90*(4), pp. 293–315.

_____. "Rational Choice and the Framing of Decisions." *Journal of Business*, October 1986, *59*(4), pp. S251–78.

- . "Loss Aversion in Riskless Choice: A Reference-Dependent Model." *Quarterly Journal of Economics*, November 1991, *106*(4), pp. 1039–61.
- . "Advances in Prospect Theory: Cumulative Representation of Uncertainty." *Journal of Risk and Uncertainty*, October 1992, 5(4), pp. 297–323.
- Tversky, Amos and Redelmeier, Donald A. "On the Framing of Multiple Prospects." *Psychological Science*, May 1992, *3*(3), pp. 191–93.
- Wilson, Timothy D. Strangers to ourselves: Discovering the adaptive unconscious. Cambridge, MA: Harvard University Press, 2002.
- Wilson, Timothy D. and Schooler, Jonathan W. "Thinking Too Much: Introspection Can Reduce the Quality of Preferences and Decisions." *Journal of Personality and Social Psychology*, February 1991, 60(2), pp. 181– 92.
- Zajonc, Robert B. "Emotions," in Daniel T. Gilbert, Susan T. Fiske, and Gardner Lindzey, eds., *Handbook of social psychology*, 4th Ed., Vol. 1. New York: Oxford University Press, 1998, pp. 591–632.

This article has been cited by:

- 1. Donncha Marron. 2014. "Informed, educated and more confident": financial capability and the problematization of personal finance consumption. *Consumption Markets & Culture* 17, 491-511. [CrossRef]
- 2. Long Wang, Chen-Bo Zhong, J. Keith Murnighan. 2014. The social and ethical consequences of a calculative mindset. *Organizational Behavior and Human Decision Processes* 125, 39-49. [CrossRef]
- 3. Christoph Merkle, Martin Weber. 2014. Do investors put their money where their mouth is? Stock market expectations and investing behavior. *Journal of Banking & Finance* 46, 372-386. [CrossRef]
- 4. Elizabeth V. Hobman, Elisha R. Frederiks. 2014. Barriers to green electricity subscription in Australia: "Love the environment, love renewable energy ... but why should I pay more?". *Energy Research & Social Science* **3**, 78-88. [CrossRef]
- 5. Dina A. Saad, Tarek Hegazy. 2014. Behavioral Economic Concepts for Funding Infrastructure Rehabilitation. *Journal of Management in Engineering* 04014089. [CrossRef]
- Huang-Ming Chang, Leonid Ivonin, Marta Diaz, Andreu Catala, Wei Chen, Matthias Rauterberg. 2014. Enacting archetypes in movies: grounding the unconscious mind in emotion-driven media. *Digital Creativity* 1-20. [CrossRef]
- 7. Yakir Levin, Itzhak Aharon. 2014. Emotion, utility maximization, and ecological rationality. *Mind* & Society. [CrossRef]
- 8. Julie A. Nelson. 2014. The power of stereotyping and confirmation bias to overwhelm accurate assessment: the case of economics, gender, and risk aversion. *Journal of Economic Methodology* 1-21. [CrossRef]
- 9. Amitai Etzioni. 2014. Treating Rationality as a Continuous Variable. Society 51, 393-400. [CrossRef]
- 10. Florian Lange, Carolin Brückner, Birte Kröger, Johannes Beller, Frank Eggert. 2014. Wasting ways: Perceived distance to the recycling facilities predicts pro-environmental behavior. *Resources, Conservation and Recycling*. [CrossRef]
- 11. Dugald Tinch, Sergio Colombo, Nick Hanley. 2014. The Impacts of Elicitation Context on Stated Preferences for Agricultural Landscapes. *Journal of Agricultural Economics* n/a-n/a. [CrossRef]
- 12. Ayala Arad. 2014. Avoiding Greedy Behavior in Situations of Uncertainty The Role of Magical Thinking. *Journal of Behavioral and Experimental Economics*. [CrossRef]
- 13. Qiuhui Pan, Lingxiao Wang, Rongrong Shi, Huan Wang, Mingfeng He. 2014. Spatial modes of cooperation based on bounded rationality. *Physica A: Statistical Mechanics and its Applications*. [CrossRef]
- 14. William Tankersley. 2014. Evolving Evolutionary Economics. *Public Administration Review* n/a-n/a. [CrossRef]
- 15. M. Cremers, A. Pareek. 2014. Short-Term Trading and Stock Return Anomalies: Momentum, Reversal, and Share Issuance. *Review of Finance*. [CrossRef]
- 16. Milan Zafirovski. 2014. Rational Choice Requiem: The Decline of an Economic Paradigm and its Implications for Sociology. *The American Sociologist*. [CrossRef]
- 17. Laura Camfield, Lucio Esposito. 2014. A Cross-Country Analysis of Perceived Economic Status and Life Satisfaction in High- and Low-Income Countries. *World Development* 59, 212-223. [CrossRef]
- Martin Binder. 2014. Should evolutionary economists embrace libertarian paternalism?. Journal of Evolutionary Economics 24, 515-539. [CrossRef]
- 19. Jason R. W. Merrick, Philip Leclerc. 2014. Modeling Adversaries in Counterterrorism Decisions Using Prospect Theory. *Risk Analysis* n/a-n/a. [CrossRef]

- 20. Xiaohua Yu, Binjian Yan, Zhifeng Gao. 2014. Can willingness-to-pay values be manipulated? Evidence from an organic food experiment in China. *Agricultural Economics* n/a-n/a. [CrossRef]
- 21. J Riis. 2014. Opportunities and barriers for smaller portions in food service: lessons from marketing and behavioral economics. *International Journal of Obesity* **38**, S19-S24. [CrossRef]
- 22. Stijn Neuteleers, Bart Engelen. 2014. Talking money: How market-based valuation can undermine environmental protection. *Ecological Economics*. [CrossRef]
- Louise Connell, Dermot Lynott. 2014. Principles of Representation: Why You Can't Represent the Same Concept Twice. *Topics in Cognitive Science* 6:10.1111/tops.2014.6.issue-3, 390-406. [CrossRef]
- 24. Jakob Kapeller, Bernhard Schütz. 2014. Debt, boom, bust: a theory of Minsky-Veblen cycles. *Journal of Post Keynesian Economics* 36, 781-814. [CrossRef]
- 25. Lisbet Berg. 2014. Who benefits from behavioural economics?. *Economic Analysis and Policy* 44, 221-232. [CrossRef]
- 26. Thomas H. B. FitzGerald, Raymond J. Dolan, Karl J. Friston. 2014. Model averaging, optimal inference, and habit formation. *Frontiers in Human Neuroscience* 8. . [CrossRef]
- 27. Risk Acceptability Analysis Based on Quantitative Psychology of Intuitive Judgments 215-244. [CrossRef]
- 28. ULRICH WITT. 2014. The future of evolutionary economics: why the modalities of explanation matter. *Journal of Institutional Economics* 1-20. [CrossRef]
- 29. S. Banerjee, F. P. de Vries, N. Hanley, D. P. van Soest. 2014. The Impact of Information Provision on Agglomeration Bonus Performance: An Experimental Study on Local Networks. *American Journal of Agricultural Economics*. [CrossRef]
- Norbert Hirschauer, Mira Lehberger, Oliver Musshoff. 2014. Happiness and Utility in Economic Thought—Or: What Can We Learn from Happiness Research for Public Policy Analysis and Public Policy Making?. Social Indicators Research. [CrossRef]
- 31. JAKOB KAPELLER, STEFAN STEINERBERGER. 2014. Modeling the evolution of preferences: an answer to Schubert and Cordes. *Journal of Institutional Economics* **10**:02, 337-347. [CrossRef]
- 32. Oliver B. Büttner, Arnd Florack, Benjamin G. Serfas. 2014. A Dual-Step and Dual-Process Model of Advertising Effects: Implications for Reducing the Negative Impact of Advertising on Children's Consumption Behaviour. *Journal of Consumer Policy* **37**:2, 161-182. [CrossRef]
- 33. Sudipt Roy, Tat Chan, Amar Cheema. 2014. Price Expectations and Purchase Decisions: Evidence from an Online Store Experiment. *Customer Needs and Solutions* 1:2, 117-130. [CrossRef]
- 34. Antonio Mastrogiorgio, Enrico Petracca. 2014. Numerals as triggers of System 1 and System 2 in the 'bat and ball' problem. *Mind & Society* 13:1, 135-148. [CrossRef]
- 35. Jonathan M. Hernandez, Athanasios Tsalatsanis, Leigh Ann Humphries, Branko Miladinovic, Benjamin Djulbegovic, Vic Velanovich. 2014. Defining Optimum Treatment of Patients With Pancreatic Adenocarcinoma Using Regret-Based Decision Curve Analysis. *Annals of Surgery* 259:6, 1208-1214. [CrossRef]
- 36. Scott C. Blum, Roxane Cohen Silver, Michael J. Poulin. 2014. Perceiving Risk in a Dangerous World: Associations between Life Experiences and Risk Perceptions. *Social Cognition* 32:3, 297-314. [CrossRef]
- 37. G. V. Lehecka, X. Wang, P. Garcia. 2014. Gone in Ten Minutes: Intraday Evidence of Announcement Effects in the Electronic Corn Futures Market. *Applied Economic Perspectives and Policy*. [CrossRef]
- 38. Ioanna D Constantiou, Christiane Lehrer, Thomas Hess. 2014. Changing information retrieval behaviours: an empirical investigation of users' cognitive processes in the choice of location-based services. *European Journal of Information Systems*. [CrossRef]

- 39. Cecilia Hyunjung Mo. 2014. The Consequences of Explicit and Implicit Gender Attitudes and Candidate Quality in the Calculations of Voters. *Political Behavior*. [CrossRef]
- 40. Alex Dubov. 2014. Moral Justification of Phase 1 Oncology Trials. *Journal of Pain and Palliative Care Pharmacotherapy* 140512054447004. [CrossRef]
- 41. Marek Palasinski. 2014. Tackling cyber-terrorism: Balancing surveillance with countercommunication. *Security Journal*. [CrossRef]
- 42. Christine Trampusch. 2014. Why preferences and institutions change: A systematic process analysis of credit rating in Germany. *European Journal of Political Research* 53:2, 328-344. [CrossRef]
- 43. W. Dixon, D. Wilson. 2014. Political economy and the social disciplines: the modern life of Das Adam Smith Problem. *Cambridge Journal of Economics* **38**:3, 623-641. [CrossRef]
- 44. Bernadette Förster, Heiko von der Gracht. 2014. Assessing Delphi panel composition for strategic foresight A comparison of panels based on company-internal and external participants. *Technological Forecasting and Social Change* 84, 215-229. [CrossRef]
- 45. Rosario Macera. 2014. Dynamic beliefs. Games and Economic Behavior . [CrossRef]
- 46. Jérôme Hergueux, Nicolas Jacquemet. 2014. Social preferences in the online laboratory: a randomized experiment. *Experimental Economics*. [CrossRef]
- OLIVIER BRETTE, THOMAS BUHLER, NATHALIE LAZARIC, KEVIN MARECHAL. 2014. Reconsidering the nature and effects of habits in urban transportation behavior. *Journal of Institutional Economics* 1-28. [CrossRef]
- 48. Jonathan F. Schulz, Urs Fischbacher, Christian Thöni, Verena Utikal. 2014. Affect and fairness: Dictator games under cognitive load. *Journal of Economic Psychology* 41, 77-87. [CrossRef]
- 49. Hendrik P. van Dalen, Kène Henkens. 2014. Comparing the effects of defaults in organ donation systems. *Social Science & Medicine* 106, 137-142. [CrossRef]
- 50. Raymond C. Sinclair, Thomas R. Cunningham. 2014. Safety activities in small businesses. *Safety Science* 64, 32-38. [CrossRef]
- Randall K. Minas, Robert F. Potter, Alan R. Dennis, Valerie Bartelt, Soyoung Bae. 2014. Putting on the Thinking Cap: Using NeuroIS to Understand Information Processing Biases in Virtual Teams. *Journal of Management Information Systems* 30:4, 49-82. [CrossRef]
- 52. Jose Ripper Kos, Bruna Mayer de Souza. 2014. Educating home users through a solar house: The Ekó House experience. *Energy and Buildings*. [CrossRef]
- 53. Juan Carlos Olabe, Xabier Basogain, Miguel Angel Olabe, Inmaculada Maíz, Carlos Castaño. 2014. Solving math and science problems in the real world with a computational mind. *Journal of New Approaches in Educational Research*. [CrossRef]
- 54. Tobias Brosch, Martin K. Patel, David Sander. 2014. Affective Influences on Energy-Related Decisions and Behaviors. *Frontiers in Energy Research* 2. [CrossRef]
- 55. Yu Henry Xie, Taewon Suh. 2014. Perceived resource deficiency and internationalization of smalland medium-sized firms. *Journal of International Entrepreneurship*. [CrossRef]
- 56. P. J. Liu, J. Wisdom, C. A. Roberto, L. J. Liu, P. A. Ubel. 2014. Using Behavioral Economics to Design More Effective Food Policies to Address Obesity. *Applied Economic Perspectives and Policy* 36:1, 6-24. [CrossRef]
- 57. A. Hamoudi, J. B. Dowd. 2014. Housing Wealth, Psychological Well-being, and Cognitive Functioning of Older Americans. *The Journals of Gerontology Series B: Psychological Sciences and Social Sciences* 69:2, 253-262. [CrossRef]

- 58. Jaap Daalhuizen, Oscar Person, Valentin Gattol. 2014. A personal matter? An investigation of students' design process experiences when using a heuristic or a systematic method. *Design Studies* 35:2, 133-159. [CrossRef]
- 59. M. M. Galizzi. 2014. What Is Really Behavioral in Behavioral Health Policy? And Does It Work?. *Applied Economic Perspectives and Policy* **36**:1, 25-60. [CrossRef]
- 60. Darla K. Munroe, Kendra McSweeney, Jeffrey L. Olson, Becky Mansfield. 2014. Using economic geography to reinvigorate land-change science. *Geoforum* 52, 12-21. [CrossRef]
- 61. James Nebus, Kah Hin Chai. 2014. Putting the "psychic" Back in Psychic Distance: Awareness, Perceptions, and Understanding as Dimensions of Psychic Distance. *Journal of International Management* 20:1, 8-24. [CrossRef]
- David M. Carballo, Paul Roscoe, Gary M. Feinman. 2014. Cooperation and Collective Action in the Cultural Evolution of Complex Societies. *Journal of Archaeological Method and Theory* 21:1, 98-133. [CrossRef]
- 63. C. Niza, C. Rudisill, P. Dolan. 2014. Vouchers versus Lotteries: What Works Best in Promoting Chlamydia Screening? A Cluster Randomized Controlled Trial. *Applied Economic Perspectives and Policy* 36:1, 109-124. [CrossRef]
- 64. Constantinos Hadjichristidis, Barbara Summers, Kevin Thomas. 2014. Unpacking estimates of task duration: The role of typicality and temporality. *Journal of Experimental Social Psychology* **51**, 45-50. [CrossRef]
- 65. Adrienne E. Eaton, Sean E. Rogers, Tracy F. H. Chang, Paula B. Voos. 2014. Choosing union representation: the role of attitudes and emotions. *Industrial Relations Journal* 45:2, 169-188. [CrossRef]
- 66. Morris AltmanBehavioral Economics, Thinking Processes, Decision Making, and Investment Behavior 43-61. [CrossRef]
- 67. Anita Kim. 2014. The Curious Case of Self-Interest: Inconsistent Effects and Ambivalence toward a Widely Accepted Construct. *Journal for the Theory of Social Behaviour* 44:1, 99-122. [CrossRef]
- 68. Massimo Peri, Daniela A. Vandone, Lucia Baldi. 2014. Internet, noise trading and commodity futures prices. *International Review of Economics & Finance*. [CrossRef]
- 69. Hans-Rüdiger Pfister, Gisela Böhm. 2014. Independent decisions are fictional from a psychological perspective. *Behavioral and Brain Sciences* 37:01, 95-96. [CrossRef]
- 70. Alexander Peysakhovich. 2014. How to Commit (If You Must): Commitment Contracts and the Dual-Self Model. *Journal of Economic Behavior & Organization*. [CrossRef]
- 71. IRIS STAUB-KAMINSKI, ANNE ZIMMER, MICHAEL JAKOB, ROBERT MARSCHINSKI. 2014. CLIMATE POLICY IN PRACTICE: A TYPOLOGY OF OBSTACLES AND IMPLICATIONS FOR INTEGRATED ASSESSMENT MODELING. *Climate Change Economics* 05:01, 1440004. [CrossRef]
- 72. R. Alexander Bentley, Michael J. O'Brien, William A. Brock. 2014. Mapping collective behavior in the big-data era. *Behavioral and Brain Sciences* **37**:01, 63-76. [CrossRef]
- 73. Bart Kamphorst, Annemarie Kalis. 2014. Why option generation matters for the design of autonomous e-coaching systems. *AI & SOCIETY*. [CrossRef]
- 74. A. Nicita, M. Rizzolli. 2014. In Dubio Pro Reo. Behavioral Explanations of Pro-defendant Bias in Procedures. *CESifo Economic Studies*. [CrossRef]
- 75. Gastón Ares, Franco Mawad, Ana Giménez, Alejandro Maiche. 2014. Influence of rational and intuitive thinking styles on food choice: Preliminary evidence from an eye-tracking study with yogurt labels. *Food Quality and Preference* **31**, 28-37. [CrossRef]

- 76. Jerome Schafer. 2014. European Commission Officials' Policy Attitudes. *JCMS: Journal of Common Market Studies* n/a-n/a. [CrossRef]
- 77. M. Lyn Exum, Diana Bailey, Eric L. Wright. 2014. False positive and false negative rates in selfreported intentions to offend: A replication and extension. *Journal of Criminal Justice* 42:1, 1-9. [CrossRef]
- 78. References 595-653. [CrossRef]
- 79. Kelvin Balcombe, Iain Fraser, Eugene McSorley. 2014. VISUAL ATTENTION AND ATTRIBUTE ATTENDANCE IN MULTI-ATTRIBUTE CHOICE EXPERIMENTS. *Journal of Applied Econometrics* n/a-n/a. [CrossRef]
- 80. Gordon L. Clark. 2014. Roepke Lecture in Economic Geography-Financial Literacy in Context. *Economic Geography* **90**:1, 1-23. [CrossRef]
- 81. Benjamin Djulbegovic, Shira Elqayam, Tea Reljic, Iztok Hozo, Branko Miladinovic, Athanasios Tsalatsanis, Ambuj Kumar, Jason Beckstead, Stephanie Taylor, Janice Cannon-Bowers. 2014. How do physicians decide to treat: an empirical evaluation of the threshold model. *BMC Medical Informatics and Decision Making* 14:1, 47. [CrossRef]
- 82. Pedro A Ortega, Daniel A Braun. 2014. Generalized Thompson sampling for sequential decisionmaking and causal inference. *Complex Adaptive Systems Modeling* 2:1, 2. [CrossRef]
- 83. Germano Resconi, Chris Hinde. 2014. Agents and rough sets. *International Journal of Computational Intelligence Systems* **7**:sup1, 137-157. [CrossRef]
- 84. Yakov Ben-Haim. 2014. Robust satisficing and the probability of survival. *International Journal of Systems Science* **45**:1, 3-19. [CrossRef]
- 85. Catherine EckelEconomic Games for Social Scientists 335-355. [CrossRef]
- 86. Caroline Brock, Bradford Barham. 2013. 'Milk is Milk': Organic Dairy Adoption Decisions and Bounded Rationality. *Sustainability* 5:12, 5416-5441. [CrossRef]
- Michael M. Atkinson. 2013. Policy, Politics and Political Science. *Canadian Journal of Political Science* 46:04, 751-772. [CrossRef]
- 88. Raj Aggarwal. 2013. Animal spirits in financial economics: A review of deviations from economic rationality. *International Review of Financial Analysis*. [CrossRef]
- M. Lindeman, A. M. Svedholm, T. Riekki, T. Raij, R. Hari. 2013. Is it just a brick wall or a sign from the universe? An fMRI study of supernatural believers and skeptics. *Social Cognitive and Affective Neuroscience* 8:8, 943-949. [CrossRef]
- 90. W.J. Wouter Botzen, Joop de Boer, Teun Terpstra. 2013. Framing of risk and preferences for annual and multi-year flood insurance. *Journal of Economic Psychology* **39**, 357-375. [CrossRef]
- 91. Jeff Vass. 2013. Restoring social creativity to immoderate publics: the case of the financially incontinent citizen. *The Sociological Review* **61**, 79-99. [CrossRef]
- 92. Lisa Holper, Ryan O. Murphy. 2013. Hemodynamic and affective correlates assessed during performance on the Columbia Card Task (CCT). *Brain Imaging and Behavior*. [CrossRef]
- Rolf Wüstenhagen, Emanuela MenichettiThe Influence of Energy Policy on Strategic Choices for Renewable Energy Investment 373-387. [CrossRef]
- 94. Conrad Heilmann. 2013. Success conditions for nudges: a methodological critique of libertarian paternalism. *European Journal for Philosophy of Science*. [CrossRef]
- 95. Anja Achtziger, Carlos Alós-Ferrer. 2013. Fast or Rational? A Response-Times Study of Bayesian Updating. *Management Science* 131105054351001. [CrossRef]

- 96. E. Florin, D. Muller, J. Pfeifer, M. T. Barbe, G. R. Fink, L. Timmermann. 2013. Subthalamic stimulation modulates self-estimation of patients with Parkinson's disease and induces risk-seeking behaviour. *Brain* 136:11, 3271-3281. [CrossRef]
- Heather J. Gordon, Evangelia Demerouti, Tanja Bipp, Pascale M. Le Blanc. 2013. The Job Demands and Resources Decision Making (JD-R-DM) Model. *European Journal of Work and Organizational Psychology* 1-15. [CrossRef]
- 98. Yijing Wang, Guido Berens, Cees B M van Riel. 2013. Competing in the Capital Market with a Good Reputation. *Corporate Reputation Review* 15:3, 198-221. [CrossRef]
- 99. J. Mehta. 2013. The discourse of bounded rationality in academic and policy arenas: pathologising the errant consumer. *Cambridge Journal of Economics* **37**:6, 1243-1261. [CrossRef]
- 100. Henrik Andersson, Mikael Svensson. 2013. Scale sensitivity and question order in the contingent valuation method. *Journal of Environmental Planning and Management* 1-17. [CrossRef]
- 101. Michael Ehrmann, Marcel Fratzscher. 2013. Dispersed communication by central bank committees and the predictability of monetary policy decisions. *Public Choice* **157**:1-2, 223-244. [CrossRef]
- 102. Santosh Anagol, Keith Jacks Gamble. 2013. Does Presenting Investment Results Asset by Asset Lower Risk Taking?. *Journal of Behavioral Finance* 14:4, 276-300. [CrossRef]
- 103. Olivier Mesly. 2013. Twenty Recommendations to Reduce Financial Predation and Blind Trust in Small Family-Owned Financial Businesses. *The Journal of Wealth Management* 16:3, 122-140. [CrossRef]
- 104. Agnes Virlics. 2013. Emotions in Economic Decision Making: A Multidisciplinary Approach. Procedia - Social and Behavioral Sciences 92, 1011-1015. [CrossRef]
- 105. Tanja Ažderska, Borka Jerman-Blažič. 2013. A Holistic Approach for Designing Human-Centric Trust Systems. *Systemic Practice and Action Research* 26:5, 417-450. [CrossRef]
- 106. Haley Engelberth, Mario F. Teisl, Eric Frohmberg, Karyn Butts, Kathleen P. Bell, Sue Stableford, Andrew E. Smith. 2013. Can fish consumption advisories do better? Providing benefit and risk information to increase knowledge. *Environmental Research* **126**, 232-239. [CrossRef]
- 107. Marius F. Niculescu, D. J. Wu. 2013. Economics of Free Under Perpetual Licensing: Implications for the Software Industry. *Information Systems Research* 140304122235009. [CrossRef]
- 108. Paul M. Peeters. 2013. Developing a long-term global tourism transport model using a behavioural approach: implications for sustainable tourism policy making. *Journal of Sustainable Tourism* 21:7, 1049-1069. [CrossRef]
- 109. Zachary Brown, Nick Johnstone, Ivan Haščič, Laura Vong, Francis Barascud. 2013. Testing the effect of defaults on the thermostat settings of OECD employees. *Energy Economics* **39**, 128-134. [CrossRef]
- 110. Marian V. Jones, Lucrezia Casulli. 2013. International Entrepreneurship: Exploring the Logic and Utility of Individual Experience Through Comparative Reasoning Approaches. *Entrepreneurship Theory and Practice* n/a-n/a. [CrossRef]
- 111. Yan Li, Liyan Yang. 2013. Asset-Pricing Implications of Dividend Volatility. *Management Science* 59:9, 2036-2055. [CrossRef]
- 112. Andrew Healy, Gabriel S. Lenz. 2013. Substituting the End for the Whole: Why Voters Respond Primarily to the Election-Year Economy. *American Journal of Political Science* n/a-n/a. [CrossRef]
- 113. Eduardo B. Andrade, Danny P. Claro, Gazi Islam. 2013. Misestimating Betting Behavior: The Role of Negative Asymmetries in Emotional Self Prediction. *Journal of Gambling Studies*. [CrossRef]
- 114. Ti-Ching Peng. 2013. A microstructural analysis of housing renovation decisions in Brisbane, Australia. *New Zealand Economic Papers* 47:2, 158-187. [CrossRef]

- 115. Byron J. Pike, Mary B. Curtis, Lawrence Chui. 2013. How Does an Initial Expectation Bias Influence Auditors' Application and Performance of Analytical Procedures?. *The Accounting Review* 88:4, 1413-1431. [CrossRef]
- 116. Luca A. Panzone. 2013. Saving money vs investing money: Do energy ratings influence consumer demand for energy efficient goods?. *Energy Economics* 38, 51-63. [CrossRef]
- 117. Calum G Turvey, Xin Gao, Rong Nie, Linping Wang, Rong Kong. 2013. Subjective Risks, Objective Risks and the Crop Insurance Problem in Rural China. *The Geneva Papers on Risk and Insurance Issues and Practice* 38:3, 612-633. [CrossRef]
- 118. Mikhail Anufriev, Cars H. Hommes, Raoul H. S. Philipse. 2013. Evolutionary selection of expectations in positive and negative feedback markets. *Journal of Evolutionary Economics* 23:3, 663-688. [CrossRef]
- 119. Peijie Wang. 2013. Reverse shooting of exchange rates. Economic Modelling 33, 71-76. [CrossRef]
- 120. Gordon L. Clark. 2013. MAPPING FINANCIAL LITERACY: COGNITION AND THE ENVIRONMENT. Geografiska Annaler: Series B, Human Geography 95:2, 131-145. [CrossRef]
- 121. Ravi Dhar, Margarita Gorlin. 2013. A dual-system framework to understand preference construction processes in choice. *Journal of Consumer Psychology*. [CrossRef]
- 122. Harmen de Weerd, Rineke Verbrugge, Bart Verheij. 2013. How much does it help to know what she knows you know? An agent-based simulation study. *Artificial Intelligence* **199-200**, 67-92. [CrossRef]
- 123. Johann Graf Lambsdorff, Manuel Schubert, Marcus Giamattei. 2013. On the role of heuristics —Experimental evidence on inflation dynamics. *Journal of Economic Dynamics and Control* 37:6, 1213-1229. [CrossRef]
- 124. A. T. Hayashi. 2013. Occasionally Libertarian: Experimental Evidence of Self-Serving Omission Bias. Journal of Law, Economics, and Organization 29:3, 711-733. [CrossRef]
- 125. Andreas Schotter, Paul W Beamish. 2013. The hassle factor: An explanation for managerial location shunning. *Journal of International Business Studies* 44:5, 521-544. [CrossRef]
- 126. Chen Li, Zhihua Li, Peter P. Wakker. 2013. If nudge cannot be applied: a litmus test of the readers' stance on paternalism. *Theory and Decision*. [CrossRef]
- 127. Andrew Healy, Neil Malhotra. 2013. Retrospective Voting Reconsidered. Annual Review of Political Science 16, 285-306. [CrossRef]
- 128. Antoine Beretti, Charles Figuières, Gilles Grolleau. 2013. Behavioral innovations: The missing capital in sustainable development?. *Ecological Economics* **89**, 187-195. [CrossRef]
- 129. Katherine D. Arbuthnott, Brett Dolter. 2013. Escalation of commitment to fossil fuels. *Ecological Economics* **89**, 7-13. [CrossRef]
- Chien Sheng Richard Chan, Haemin Dennis Park. 2013. The influence of dispositional affect and cognition on venture investment portfolio concentration. *Journal of Business Venturing* 28:3, 397-412. [CrossRef]
- 131. Marco Grasso. 2013. Climate ethics: with a little help from moral cognitive neuroscience. *Environmental Politics* 22, 377-393. [CrossRef]
- 132. Paul SJ Miller, Elyce A Biddle, Johanna Maria van Dongen, Maurits W van Tulder, Emile Tompa, Ian Shemilt, Paul SJ MillerEconomic incentives to enhance safety behaviour in workers for preventing occupational injuries. [CrossRef]
- 133. Amar Hamoudi, Jennifer Beam Dowd. 2013. Physical Health Effects of the Housing Boom: Quasi-Experimental Evidence From the Health and Retirement Study. *American Journal of Public Health* e1e7. [CrossRef]

- 134. R. Kent Weaver. 2013. Compliance Regimes and Barriers to Behavioral Change. *Governance* n/a-n/ a. [CrossRef]
- 135. James J. Chrisman, Esra Memili, Kaustav Misra. 2013. Nonfamily Managers, Family Firms, and the Winner's Curse: The Influence of Noneconomic Goals and Bounded Rationality. *Entrepreneurship Theory and Practice* n/a-n/a. [CrossRef]
- 136. Neil Malhotra, Yotam Margalit, Cecilia Hyunjung Mo. 2013. Economic Explanations for Opposition to Immigration: Distinguishing between Prevalence and Conditional Impact. *American Journal of Political Science* 57:2, 391-410. [CrossRef]
- 137. Gerrit Antonides, Leonie Cramer. 2013. Impact of limited cognitive capacity and feelings of guilt and excuse on the endowment effects for hedonic and utilitarian types of foods. *Appetite* . [CrossRef]
- 138. P. A. Ortega, D. A. Braun. 2013. Thermodynamics as a theory of decision-making with informationprocessing costs. *Proceedings of the Royal Society A: Mathematical, Physical and Engineering Sciences* 469:2153, 20120683-20120683. [CrossRef]
- Robert Apel. 2013. Sanctions, Perceptions, and Crime: Implications for Criminal Deterrence. *Journal of Quantitative Criminology* 29:1, 67-101. [CrossRef]
- 140. Andrea Masini, Emanuela Menichetti. 2013. Investment decisions in the renewable energy sector: An analysis of non-financial drivers. *Technological Forecasting and Social Change* **80**:3, 510-524. [CrossRef]
- 141. L. R. Skov, S. Lourenço, G. L. Hansen, B. E. Mikkelsen, C. Schofield. 2013. Choice architecture as a means to change eating behaviour in self-service settings: a systematic review. *Obesity Reviews* 14:3, 187-196. [CrossRef]
- 142. Rong-Chang Jou, Ke-Hong Chen. 2013. An application of cumulative prospect theory to freeway drivers' route choice behaviours. *Transportation Research Part A: Policy and Practice* 49, 123-131. [CrossRef]
- 143. Helen Suich. 2013. The effectiveness of economic incentives for sustaining community based natural resource management. *Land Use Policy* **31**, 441-449. [CrossRef]
- 144. Tryggvi Thorgeirsson, Ichiro Kawachi. 2013. Behavioral Economics. American Journal of Preventive Medicine 44:2, 185-189. [CrossRef]
- 145. Raimo P. Hämäläinen, Jukka Luoma, Esa Saarinen. 2013. On the importance of behavioral operational research: The case of understanding and communicating about dynamic systems. *European Journal of Operational Research*. [CrossRef]
- 146. Carolyn H. Declerck, Christophe Boone, Griet Emonds. 2013. When do people cooperate? The neuroeconomics of prosocial decision making. *Brain and Cognition* **81**, 95-117. [CrossRef]
- 147. Michel J.J. Handgraaf, Margriet A. Van Lidth de Jeude, Kirstin C. Appelt. 2013. Public praise vs. private pay: Effects of rewards on energy conservation in the workplace. *Ecological Economics* 86, 86-92. [CrossRef]
- 148. Scott Kelly, Michelle Shipworth, David Shipworth, Michael Gentry, Andrew Wright, Michael Pollitt, Doug Crawford-Brown, Kevin Lomas. 2013. Predicting the diversity of internal temperatures from the English residential sector using panel methods. *Applied Energy* **102**, 601-621. [CrossRef]
- 149. Sumitava Mukherjee, Narayanan SrinivasanAttention in preferential choice 117-134. [CrossRef]
- 150. Malcolm Baker, Jeffrey WurglerBehavioral Corporate Finance: An Updated Survey 357-424. [CrossRef]
- 151. Thomas BRUDERMANN, Romana RAUTER, Yoshiki YAMAGATA. 2013. Behavioral Aspects of Urban Resilience. *Innovation and Supply Chain Management* 7:3, 83. [CrossRef]
- 152. Peter T. Radu, Samuel M. McClureNeuroeconomics and Addiction 413-423. [CrossRef]

- 153. Sanjit Dhami, Ali al-Nowaihi. 2013. An extension of the Becker proposition to non-expected utility theory. *Mathematical Social Sciences* 65:1, 10-20. [CrossRef]
- 154. N.P. Nguyen, J.S. Shortle, P.M. Reed, T.T. Nguyen. 2013. Water quality trading with asymmetric information, uncertainty and transaction costs: A stochastic agent-based simulation. *Resource and Energy Economics* 35:1, 60-90. [CrossRef]
- 155. J.A. Nelson. 2013. Ethics and the economist: What climate change demands of us. *Ecological Economics* 85, 145-154. [CrossRef]
- 156. T. W. Fawcett, S. Hamblin, L.-A. Giraldeau. 2013. Exposing the behavioral gambit: the evolution of learning and decision rules. *Behavioral Ecology* 24:1, 2-11. [CrossRef]
- 157. Krishna P. Miyapuram, V.S. Chandrasekhar PammiUnderstanding decision neuroscience: A multidisciplinary perspective and neural substrates 239-266. [CrossRef]
- 158. Alexander Kempf, Christoph Merkle, Alexandra Niessen-Ruenzi. 2013. Low Risk and High Return
 Affective Attitudes and Stock Market Expectations. *European Financial Management* no-no. [CrossRef]
- 159. Karl-Jacob Mickelsson. 2013. Customer activity in service. *Journal of Service Management* 24:5, 534-552. [CrossRef]
- 160. Dominik F. Riedl, Lutz Kaufmann, Carsten Zimmermann, Johan L. Perols. 2013. Reducing uncertainty in supplier selection decisions: Antecedents and outcomes of procedural rationality. *Journal* of Operations Management 31:1-2, 24-36. [CrossRef]
- 161. Ioanna Constantiou, Natascha Hoebel, Roberto V. Zicari. 2012. How do framing strategies influence the user's choice of content on the Web?. *Concurrency and Computation: Practice and Experience* 24:17, 2207–2220. [CrossRef]
- 162. Jeremy J Clark, Nick G Hollon, Paul EM Phillips. 2012. Pavlovian valuation systems in learning and decision making. *Current Opinion in Neurobiology* 22:6, 1054-1061. [CrossRef]
- 163. Stephen Carpenter, Kenneth Arrow, Scott Barrett, Reinette Biggs, William Brock, Anne-Sophie Crépin, Gustav Engström, Carl Folke, Terry Hughes, Nils Kautsky, Chuan-Zhong Li, Geoffrey McCarney, Kyle Meng, Karl-Göran Mäler, Stephen Polasky, Marten Scheffer, Jason Shogren, Thomas Sterner, Jeffrey Vincent, Brian Walker, Anastasios Xepapadeas, Aart Zeeuw. 2012. General Resilience to Cope with Extreme Events. *Sustainability* 4:12, 3248-3259. [CrossRef]
- 164. Nikos L. D. Chatzisarantis, Ying Hwa Kee, Hannah Kyaw Thaung, Martin S. Hagger. 2012. When small losses do not loom larger than small gains: Effects of contextual autonomy support and goal contents on behavioural responses to small losses and small gains. *British Journal of Social Psychology* 51:4, 690-708. [CrossRef]
- 165. Jan Larsson, Inger Holmström. 2012. Understanding anesthesia training and trainees. *Current Opinion in Anaesthesiology* 25:6, 681-685. [CrossRef]
- 166. Gary E. Bolton, Axel Ockenfels, Ulrich W. Thonemann. 2012. Managers and Students as Newsvendors. *Management Science* 58:12, 2225-2233. [CrossRef]
- 167. Robert Henry Cox, Daniel Béland. 2012. Valence, Policy Ideas, and the Rise of Sustainability. Governance 1-22. [CrossRef]
- 168. Alain Samson, Benjamin G. Voyer. 2012. Two minds, three ways: dual system and dual process models in consumer psychology. AMS Review 2:2-4, 48-71. [CrossRef]
- 169. Jan Drugowitsch, Alexandre Pouget. 2012. Probabilistic vs. non-probabilistic approaches to the neurobiology of perceptual decision-making. *Current Opinion in Neurobiology* 22:6, 963-969. [CrossRef]
- 170. Jorge E. Araña, Carmelo J. León. 2012. Can Defaults Save the Climate? Evidence from a Field Experiment on Carbon Offsetting Programs. *Environmental and Resource Economics*. [CrossRef]

- 171. Mikhail Anufriev,, Cars Hommes. 2012. Evolutionary Selection of Individual Expectations and Aggregate Outcomes in Asset Pricing Experiments. *American Economic Journal: Microeconomics* 4:4, 35-64. [Abstract] [View PDF article] [PDF with links]
- 172. A. Ferscha, K. Farrahi, J. Hoven, D. Hales, A. Nowak, P. Lukowicz, D. Helbing. 2012. Socio-inspired ICT. *The European Physical Journal Special Topics* **214**:1, 401-434. [CrossRef]
- 173. Christine Parker. 2012. The war on cartels and the social meaning of deterrence. Regulation & Governance n/a-n/a. [CrossRef]
- 174. Eline van der Heijden, Tobias J. Klein, Wieland Müller, Jan Potters. 2012. Framing effects and impatience: Evidence from a large scale experiment. *Journal of Economic Behavior & Organization* 84:2, 701-711. [CrossRef]
- 175. Yu-Feng Huang, Feng-Yang Kuo. 2012. How impulsivity affects consumer decision-making in ecommerce. *Electronic Commerce Research and Applications* 11:6, 582-590. [CrossRef]
- 176. Paul De Grauwe, Pablo Rovira KaltwasserThe Exchange Rate in a Behavioral Finance Framework 111-132. [CrossRef]
- 177. Samuel B. Graves, Jeffrey L. Ringuest. 2012. Patient Decision Making: Exponential versus Hyperbolic Discounting. *Managerial and Decision Economics* **33**:7-8, 453-462. [CrossRef]
- 178. Morris Altman. 2012. Implications of behavioural economics for financial literacy and public policy. *The Journal of Socio-Economics* 41:5, 677-690. [CrossRef]
- 179. Carlos Chiapa, José Luis Garrido, Silvia Prina. 2012. The effect of social programs and exposure to professionals on the educational aspirations of the poor. *Economics of Education Review* **31**:5, 778-798. [CrossRef]
- 180. D. Wade Hands. 2012. Foundations of Contemporary Revealed Preference Theory. *Erkenntnis* . [CrossRef]
- 181. A. Achtziger, C. Alos-Ferrer, S. Hugelschafer, M. Steinhauser. 2012. The neural basis of belief updating and rational decision making. *Social Cognitive and Affective Neuroscience*. [CrossRef]
- 182. Martin Wolf, Roman Beck, Immanuel Pahlke. 2012. Mindfully resisting the bandwagon: reconceptualising IT innovation assimilation in highly turbulent environments. *Journal of Information Technology* 27:3, 213-235. [CrossRef]
- Rachael Shwom, Janet A. Lorenzen. 2012. Changing household consumption to address climate change: social scientific insights and challenges. Wiley Interdisciplinary Reviews: Climate Change 3:5, 379-395. [CrossRef]
- 184. Coralio Ballester Pla, Penélope Hernández. 2012. Bounded Rationality. *Revista Internacional de Sociología*, ahead of print. [CrossRef]
- 185. MARKUS MICHAEL GEIPEL. 2012. MODULARITY, DEPENDENCE AND CHANGE. Advances in Complex Systems 1250083. [CrossRef]
- 186. WILLIAM P. BOTTOM, DEJUN TONY KONG. 2012. "THE CASUAL CRUELTY OF OUR PREJUDICES": ON WALTER LIPPMANN'S THEORY OF STEREOTYPE AND ITS "OBLITERATION" IN PSYCHOLOGY AND SOCIAL SCIENCE. Journal of the History of the Behavioral Sciences n/a-n/a. [CrossRef]
- Leonhard K. Lades. 2012. Towards an incentive salience model of intertemporal choice. *Journal of Economic Psychology* 33:4, 833-841. [CrossRef]
- 188. Leslie E. Sekerka, Derek Stimel. 2012. Environmental sustainability decision-making: clearing a path to change. *Journal of Public Affairs* 12:3, 195-205. [CrossRef]
- 189. Juan Díez Medrano, Michael Braun. 2012. Uninformed citizens and support for free trade. *Review of International Political Economy* 19:3, 448-476. [CrossRef]

- 190. Enrico Rubaltelli, Sergio Agnoli. 2012. The emotional cost of charitable donations. Cognition & Emotion 26:5, 769-785. [CrossRef]
- 191. Keith Thomas. 2012. A structural perspective to advancing complex social outcomes. *Journal of Public Affairs* 12:3, 206-213. [CrossRef]
- P. Bordalo, N. Gennaioli, A. Shleifer. 2012. Salience Theory of Choice Under Risk. *The Quarterly Journal of Economics* 127:3, 1243-1285. [CrossRef]
- 193. David Zilberman, Jinhua Zhao, Amir Heiman. 2012. Adoption Versus Adaptation, with Emphasis on Climate Change. *Annual Review of Resource Economics* 4, 27-53. [CrossRef]
- 194. D. Malatesta. 2012. The Link between Information and Bargaining Efficiency. *Journal of Public Administration Research and Theory* 22:3, 527-551. [CrossRef]
- 195. David J.T. Sumpter, Natalia Zabzina, Stamatios C. Nicolis. 2012. Six Predictions about the Decision Making of Animal and Human Groups. *Managerial and Decision Economics* 33:5-6, 295-309. [CrossRef]
- 196. M. Elisabeth Paté-Cornell. 2012. Games, Risks, and Analytics: Several Illustrative Cases Involving National Security and Management Situations. *Decision Analysis* 9:2, 186-203. [CrossRef]
- 197. Jan P. Voon, Jan Cham Voon. 2012. A structural model of consumption: An application to China during the global financial crisis. *The Journal of Socio-Economics* 41:3, 284-288. [CrossRef]
- 198. Arie W. Kruglanski, Lauren M. Boyatzi. 2012. THE PSYCHOLOGY OF CLOSED AND OPEN MINDEDNESS, RATIONALITY, AND DEMOCRACY. *Critical Review* 24:2, 217-232. [CrossRef]
- 199. José Manuel Hurtado González, Miguel Ángel Calderón, José Luis Galán González. 2012. The alignment of managers' mental models with the balanced scorecard strategy map. *Total Quality Management & Business Excellence* 23:5-6, 613-628. [CrossRef]
- 200. Sanjay Goel, Pietro Mazzola, Phillip H. Phan, Torsten M. Pieper, Ramona K. Zachary. 2012. Strategy, ownership, governance, and socio-psychological perspectives on family businesses from around the world. *Journal of Family Business Strategy* 3:2, 54-65. [CrossRef]
- 201. Fabio Landini. 2012. Technology, property rights and organizational diversity in the software industry. *Structural Change and Economic Dynamics* 23:2, 137-150. [CrossRef]
- 202. Chikashi Tsuji. 2012. Do industries contain predictive information for the Fama-French factors?. *Quantitative Finance* 12:6, 969-991. [CrossRef]
- 203. Mikhail Anufriev, Cars Hommes. 2012. Evolution of market heuristics. *The Knowledge Engineering Review* 27:02, 255-271. [CrossRef]
- 204. Dan M. Kahan, Ellen Peters, Maggie Wittlin, Paul Slovic, Lisa Larrimore Ouellette, Donald Braman, Gregory Mandel. 2012. The polarizing impact of science literacy and numeracy on perceived climate change risks. *Nature Climate Change*. [CrossRef]
- 205. Anat Bracha, Donald J. Brown. 2012. Affective decision making: A theory of optimism bias. *Games and Economic Behavior* **75**:1, 67-80. [CrossRef]
- 206. Jordan W. Smith, Christos Siderelis, Roger L. Moore, Dorothy H. Anderson. 2012. The effects of place meanings and social capital on desired forest management outcomes: A stated preference experiment. *Landscape and Urban Planning* **106**:2, 207-218. [CrossRef]
- 207. Caspar G. Chorus. 2012. What about behaviour in travel demand modelling? An overview of recent progress. *Transportation Letters: The International Journal of Transportation Research* 4:2, 93-104. [CrossRef]
- 208. U. Bhattacharya, A. Hackethal, S. Kaesler, B. Loos, S. Meyer. 2012. Is Unbiased Financial Advice to Retail Investors Sufficient? Answers from a Large Field Study. *Review of Financial Studies* 25:4, 975-1032. [CrossRef]

- 209. Meghana A. Bhatt. 2012. Evaluation and associations: A neural-network model of advertising and consumer choice. *Journal of Economic Behavior & Organization* 82:1, 236-255. [CrossRef]
- 210. R. ALEXANDER BENTLEY, PAUL ORMEROD. 2012. ACCELERATED INNOVATION AND INCREASED SPATIAL DIVERSITY OF US POPULAR CULTURE. Advances in Complex Systems 15:01n02, 1150011. [CrossRef]
- 211. Shu-Heng Chen, Shu G. WangEmergent Complexity in Agent-Based Computational Economics 131-150. [CrossRef]
- 212. Cassey LeeBounded Rationality and the Emergence of Simplicity Amidst Complexity 111-129. [CrossRef]
- 213. Stephen Kwaku Asare, Arnold Wright. 2012. The Effect of Change in the Reporting Threshold and Type of Control Deficiency on Equity Analysts' Evaluation of the Reliability of Future Financial Statements. *AUDITING: A Journal of Practice & Theory* 120202131513008. [CrossRef]
- 214. Thomas Sturm. 2012. The "Rationality Wars" in Psychology: Where They Are and Where They Could Go. *Inquiry* 55:1, 66-81. [CrossRef]
- 215. Ross Finnie. 2012. Access to post-secondary education: The importance of culture. *Children and Youth Services Review*. [CrossRef]
- 216. Björn Frank. 2012. The formation of consumer attitudes and intentions towards fast food restaurants: How do teenagers differ from adults?. *Managing Service Quality* **22**:3, 260-282. [CrossRef]
- 217. Moshe Sniedovich. 2012. Black Swans, New Nostradamuses, Voodoo decision theories, and the science of decision making in the face of severe uncertainty. *International Transactions in Operational Research* 19:1-2, 253-281. [CrossRef]
- 218. Han Bleichrodt, Kirsten I.M. Rohde, Tom Van Ourti. 2012. An experimental test of the concentration index. *Journal of Health Economics* . [CrossRef]
- 219. Amy M. Young, Mary D. Hinesly. 2012. Identifying Millennials' key influencers from early childhood: insights into current consumer preferences. *Journal of Consumer Marketing* 29:2, 146-155. [CrossRef]
- 220. Takashi MARUYAMA, Kimio HASHIMOTO, Syungo UEDA, Masahiro NAKAGAWA. 2012. Emotional Measurements Method on the basis of Fractal Dimension of EEG -Consider the Use of Refreshing Shampoo-. *Journal of Japan Society for Fuzzy Theory and Intelligent Informatics* 24:6, 1137-1153. [CrossRef]
- 221. Mario Hayek, Michael Harvey. 2012. Attention deficit/hyperactive disorder as an entrepreneurial "marker" among family business members: A social learning perspective. *Journal of Family Business Management* 2:1, 6-22. [CrossRef]
- 222. D. Wade Hands The Positive-Normative Dichotomy and Economics 219-239. [CrossRef]
- 223. Miguel Alzola, Gary R. Weaver. 2012. Reviving Tradition. *Business Ethics Quarterly* 22:2, 377-404. [CrossRef]
- 224. Evmorfia Argyriou. 2012. Consumer Intentions to Revisit Online Retailers: A Mental Imagery Account. Psychology & Marketing 29:1, 25-35. [CrossRef]
- 225. Benjamin Djulbegovic, Iztok Hozo, Jason Beckstead, Athanasios Tsalatsanis, Stephen G Pauker. 2012. Dual processing model of medical decision-making. *BMC Medical Informatics and Decision Making* 12:1, 94. [CrossRef]
- 226. Dmitri G. Markovitch, Joel H. Steckel. 2012. Do initial stock price reactions provide a good measurement stick for marketing strategies?: The case of new product introductions in the US. *European Journal of Marketing* 46:3/4, 406-421. [CrossRef]
- 227. J&r my Celse, Jan P. Voon, Jan Cham Voon, J&rgen Drud Hansen, Hassan Molana, Catia Montagna, J&rgen Ulff-M&ller Nielsen. 2012. Is the positional bias an artefact? Distinguishing positional concerns from egalitarian concerns. *Journal of Socio-Economics* 41:3, 277. [CrossRef]

- 228. David Strutton, Jeffrey Lewin. 2012. Investigating consumers' responses to the Great Recession. Journal of Consumer Marketing 29:5, 378-388. [CrossRef]
- 229. Paul Ormerod. 2012. Terrorist networks and the lethality of attacks: an illustrative agent based model on evolutionary principles. *Security Informatics* 1:1, 16. [CrossRef]
- José Atilano Pena López. 2011. Economía, sociedad y ética: Una propuesta integrativa. Arbor 187:752, 1245-1258. [CrossRef]
- 231. Cars Hommes, Thomas Lux. 2011. INDIVIDUAL EXPECTATIONS AND AGGREGATE BEHAVIOR IN LEARNING-TO-FORECAST EXPERIMENTS. *Macroeconomic Dynamics* 1-29. [CrossRef]
- 232. Hongzhi Gao, John G. Knight, Hongxia Zhang, Damien Mather. 2011. Guilt by association: Heuristic risks for foreign brands during a product-harm crisis in China. *Journal of Business Research*. [CrossRef]
- 233. Amitai Etzioni. 2011. On Communitarian and Global Sources of Legitimacy. The Review of Politics 73:01, 105-122. [CrossRef]
- 234. Ulrich Witt. 2011. Economic Behavior—Evolutionary Versus Behavioral Perspectives. *Biological Theory* **6**:4, 388-398. [CrossRef]
- 235. David Hirshleifer, Siew Hong TeohPsychological Influences on Financial Regulation and Policy 151-167. [CrossRef]
- 236. Morris AltmanProspect Theory and Behavioral Finance 191-209. [CrossRef]
- 237. Timo Goeschl, Grischa Perino. 2011. Instrument Choice and Motivation: Evidence from a Climate Change Experiment. *Environmental and Resource Economics*. [CrossRef]
- 238. Sudipto Mundle, N.R. Bhanumurthy, Surajit Das. 2011. Fiscal consolidation with high growth: A policy simulation model for India. *Economic Modelling* 28:6, 2657-2668. [CrossRef]
- 239. Christian Reise, Oliver Musshoff, Karol Granoszewski, Achim Spiller. 2011. Which factors influence the expansion of bioenergy? An empirical study of the investment behaviours of German farmers. *Ecological Economics*. [CrossRef]
- 240. Kyu Sang Lee. 2011. Three ways of linking laboratory endeavours to the realm of policies. *The European Journal of the History of Economic Thought* 1-22. [CrossRef]
- 241. Dawn R. Elm, Tara J. Radin. 2011. Ethical Decision Making: Special or No Different?. *Journal of Business Ethics*. [CrossRef]
- 242. J. Ginges, S. Atran. 2011. War as a moral imperative (not just practical politics by other means). *Proceedings of the Royal Society B: Biological Sciences* 278:1720, 2930-2938. [CrossRef]
- 243. Stefano Fiori. 2011. Forms of Bounded Rationality: The Reception and Redefinition of Herbert A. Simon's Perspective. *Review of Political Economy* **23**:4, 587-612. [CrossRef]
- 244. Hamidreza Amouzegar, Mohammad Jafar Tarokh, Anahita Naghilouye Hidaji. 2011. A New Soa Security Model to Protect Against Web Competitive Intelligence Attacks by Software Agents. *International Journal of Information Security and Privacy* **3**:4, 18-28. [CrossRef]
- 245. P. Dolan, M. Hallsworth, D. Halpern, D. King, R. Metcalfe, I. Vlaev. 2011. Influencing behaviour: the mindspace way. *Journal of Economic Psychology* . [CrossRef]
- 246. Roger N. Jones. 2011. The latest iteration of IPCC uncertainty guidance—an author perspective. *Climatic Change*. [CrossRef]
- 247. Pier Luigi Porta. 2011. Lombard Enlightenment and Classical Political Economy. *The European Journal of the History of Economic Thought* 1-30. [CrossRef]
- 248. Shu-Heng Chen. 2011. Varieties of agents in agent-based computational economics: A historical and an interdisciplinary perspective. *Journal of Economic Dynamics and Control*. [CrossRef]

- 249. Lisa Bortolotti. 2011. Does reflection lead to wise choices?. *Philosophical Explorations* 14:3, 297-313. [CrossRef]
- 250. Jan Trzaskowski. 2011. Behavioural Economics, Neuroscience, and the Unfair Commercial Practises Directive. *Journal of Consumer Policy*. [CrossRef]
- 251. Georg Ch. Pflug, Alois Pichler, David Wozabal. 2011. The 1/N investment strategy is optimal under high model ambiguity. *Journal of Banking & Finance*. [CrossRef]
- 252. Andrea Insch, Rebecca S. Prentice, John G. Knight. 2011. Retail buyers' decision-making and buy national campaigns. *Australasian Marketing Journal (AMJ)*. [CrossRef]
- 253. Giuseppe Di Vita. 2011. Curbing Corruption in Public Administration: A Case Study from Italy. *International Journal of Public Administration* **34**:10, 631-645. [CrossRef]
- 254. Benjamin Djulbegovic. 2011. Uncertainty and Equipoise: At Interplay Between Epistemology, Decision Making and Ethics. *The American Journal of the Medical Sciences* 1. [CrossRef]
- 255. Rolf Wüstenhagen, Emanuela Menichetti. 2011. Strategic choices for renewable energy investment: Conceptual framework and opportunities for further research. *Energy Policy*. [CrossRef]
- 256. Susumu Cato. 2011. SZPILRAJN, ARROW AND SUZUMURA: CONCISE PROOFS OF EXTENSION THEOREMS AND AN EXTENSION. *Metroeconomica* no-no. [CrossRef]
- 257. A. Cigno. 2011. How to Deal with Covert Child Labor and Give Children an Effective Education, in a Poor Developing Country. *The World Bank Economic Review*. [CrossRef]
- 258. X. Gine, G. Mansuri, M. Picon. 2011. Does a Picture Paint a Thousand Words? Evidence from a Microcredit Marketing Experiment. *The World Bank Economic Review*. [CrossRef]
- 259. Cinla Akinci, Eugene Sadler-Smith. 2011. Intuition in Management Research: A Historical Review. International Journal of Management Reviews no-no. [CrossRef]
- 260. Dongjae Jung, Wooseong Jeong. 2011. Nudge: A Tool for Better Policy Impacts and Its Limitations under Various Policy Contexts. *Public Administration Review* **71**:4, 653-656. [CrossRef]
- 261. Gordana Manevska-Tasevska, Helena Hansson. 2011. Does Managerial Behavior Determine Farm Technical Efficiency? A Case of Grape Production in an Economy in Transition. *Managerial and Decision Economics* n/a-n/a. [CrossRef]
- 262. Niclas Berggren. 2011. Time for behavioral political economy? An analysis of articles in behavioral economics. *The Review of Austrian Economics*. [CrossRef]
- 263. R. A. Bentley, P. Ormerod, S. Shennan. 2011. Population-level neutral model already explains linguistic patterns. *Proceedings of the Royal Society B: Biological Sciences* 278:1713, 1770-1772. [CrossRef]
- 264. Feng Li, Dariusz Pieńkowski, Aad van Moorsel, Chris Smith. 2011. A Holistic Framework for Trust in Online Transactions. *International Journal of Management Reviews* no-no. [CrossRef]
- 265. Christof Knoeri, Claudia R. Binder, Hans-Joerg Althaus. 2011. Decisions on recycling: Construction stakeholders' decisions regarding recycled mineral construction materials. *Resources, Conservation and Recycling*. [CrossRef]
- 266. FRANCISCO FATAS-VILLAFRANCA, DULCE SAURA, FRANCISCO J. VAZQUEZ. 2011. A Dynamic Model of Public Opinion Formation. *Journal of Public Economic Theory* 13:3, 417-441. [CrossRef]
- 267. Zachary G. Arens, Roland T. Rust. 2011. The duality of decisions and the case for impulsiveness metrics. *Journal of the Academy of Marketing Science*. [CrossRef]
- 268. Amitai Etzioni. 2011. Behavioural Economics: Next Steps. Journal of Consumer Policy . [CrossRef]
- 269. Warren Bailey, Alok Kumar, David Ng. 2011. Behavioral biases of mutual fund investors. *Journal of Financial Economics*. [CrossRef]

- 270. A. Jud, P. Perrig-Chiello, P. Voll. 2011. Less effort in worsening child protection cases? The timecourse of intensity of services. *Children and Youth Services Review*. [CrossRef]
- 271. Gerrit AntonidesBehavioural Economics Applied 500-524. [CrossRef]
- 272. José J. Cañas, Boris B. Velichkovsky, Boris M. VelichkovskyHuman Factors and Ergonomics 316-337. [CrossRef]
- 273. R. A. Pollak. 2011. Family Bargaining and Taxes: A Prolegomenon to the Analysis of Joint Taxation. CESifo Economic Studies. [CrossRef]
- 274. Levon Barseghyan, Jeffrey Prince, Joshua C. Teitelbaum. 2011. Are Risk Preferences Stable across Contexts? Evidence from Insurance Data. *American Economic Review* 101:2, 591-631. [Abstract] [View PDF article] [PDF with links]
- 275. Margaret J. Eppstein, David K. Grover, Jeffrey S. Marshall, Donna M. Rizzo. 2011. An agent-based model to study market penetration of plug-in hybrid electric vehicles. *Energy Policy* . [CrossRef]
- 276. Roberta Patalano. 2011. Resistance to Change: Historical Excursus and Contemporary Interpretations. *Review of Political Economy* 23:2, 249-266. [CrossRef]
- 277. Rose Mcdermott. 2011. New Directions for Experimental Work in International Relations. International Studies Quarterly no-no. [CrossRef]
- 278. Anke Fischer, Klaus Glenk. 2011. One model fits all? On the moderating role of emotional engagement and confusion in the elicitation of preferences for climate change adaptation policies. *Ecological Economics* 70:6, 1178-1188. [CrossRef]
- 279. Andreia C. Santos, Jennifer A. Roberts, Mauricio L. Barreto, Sandy Cairncross. 2011. Demand for sanitation in Salvador, Brazil: A hybrid choice approach. *Social Science & Medicine* 72:8, 1325-1332. [CrossRef]
- 280. K. Albrecht, K. G. Volz, M. Sutter, D. I. Laibson, D. Y. von Cramon. 2011. What is for me is not for you: brain correlates of intertemporal choice for self and other. *Social Cognitive and Affective Neuroscience* 6:2, 218-225. [CrossRef]
- 281. John Livanas. 2011. Are Investors Rational and Does it Matter? Determining the Expected Utility Function for a Group of Investors. *Journal of Behavioral Finance* 12:2, 53-67. [CrossRef]
- 282. S. C. Dow. 2011. Cognition, market sentiment and financial instability. *Cambridge Journal of Economics* 35:2, 233-249. [CrossRef]
- 283. Devin Fidler. 2011. Foresight defined as a component of Strategic Management. Futures . [CrossRef]
- 284. Peter Kenning, Marc Linzmajer. 2011. Consumer neuroscience: an overview of an emerging discipline with implications for consumer policy. *Journal für Verbraucherschutz und Lebensmittelsicherheit* 6:1, 111-125. [CrossRef]
- 285. R. Alexander Bentley, Paul Ormerod, Michael Batty. 2011. Evolving social influence in large populations. *Behavioral Ecology and Sociobiology* 65:3, 537-546. [CrossRef]
- 286. Ray Paternoster, Greg Pogarsky, Gregory Zimmerman. 2011. Thoughtfully Reflective Decision Making and the Accumulation of Capital: Bringing Choice Back In. *Journal of Quantitative Criminology* 27:1, 1-26. [CrossRef]
- 287. Nabil I. Al-Najjar, Luciano de CastroSubjective Probability . [CrossRef]
- 288. John Sautter, Natalia Czap, Colby Kruse, Gary Lynne. 2011. Farmers' Decisions Regarding Carbon Sequestration: A Metaeconomic View. *Society & Natural Resources* 24:2, 133-147. [CrossRef]
- 289. A. P. Zwane, J. Zinman, E. Van Dusen, W. Pariente, C. Null, E. Miguel, M. Kremer, D. S. Karlan, R. Hornbeck, X. Gine, E. Duflo, F. Devoto, B. Crepon, A. Banerjee. 2011. Being surveyed can change later behavior and related parameter estimates. *Proceedings of the National Academy of Sciences* 108:5, 1821-1826. [CrossRef]

- 290. Michael Lamla, Sarah Lein. 2011. What matters when? The impact of ECB communication on financial market expectations. *Applied Economics* 1-21. [CrossRef]
- 291. Sacha Bourgeois-Gironde, Marianne Guille. 2011. Keynes's animal spirits vindicated: an analysis of recent empirical and neural data on money illusion. *Journal of Post Keynesian Economics* 34:2, 331-352. [CrossRef]
- 292. Rob J.F. Burton, Upananda Herath Paragahawewa. 2011. Creating culturally sustainable agrienvironmental schemes. *Journal of Rural Studies* 27:1, 95-104. [CrossRef]
- 293. Yu-feng Huang, Feng-yang Kuo. 2011. An eye-tracking investigation of internet consumers' decision deliberateness. *Internet Research* 21:5, 541-561. [CrossRef]
- 294. Leonie Cramer, Gerrit Antonides. 2011. Endowment effects for hedonic and utilitarian food products. *Food Quality and Preference* 22:1, 3-10. [CrossRef]
- 295. Eric Bentzen, John K. Christiansen, Claus J. Varnes. 2011. What attracts decision makers' attention?: Managerial allocation of time at product development portfolio meetings. *Management Decision* 49:3, 330-349. [CrossRef]
- 296. Sejin Ha, Yun Jung Lee. 2011. Determinants of consumer-driven healthcare: Self-confidence in information search, health literacy, and trust in information sources. *International Journal of Pharmaceutical and Healthcare Marketing* 5:1, 8-24. [CrossRef]
- 297. Athanasios Tsalatsanis, Laura E Barnes, Iztok Hozo, Benjamin Djulbegovic. 2011. Extensions to Regret-based Decision Curve Analysis: An application to hospice referral for terminal patients. BMC Medical Informatics and Decision Making 11:1, 77. [CrossRef]
- 298. Cassey Lee. 2011. BOUNDED RATIONALITY AND THE EMERGENCE OF SIMPLICITY AMIDST COMPLEXITY. *Journal of Economic Surveys* no-no. [CrossRef]
- 299. Benjamin Djulbegovic, Iztok Hozo, Sander GreenlandUncertainty in Clinical Medicine 299-356. [CrossRef]
- 300. Wilson Sy. 2011. Redesigning Choice and Competition in Australian Superannuation. *Rotman* International Journal of Pension Management 4:1, 52-61. [CrossRef]
- 301. Ricardo Daziano, Denis Bolduc. 2011. Incorporating pro-environmental preferences towards green automobile technologies through a Bayesian hybrid choice model. *Transportmetrica* 1-33. [CrossRef]
- 302. Fynnwin Prager, Garrett Ryan Beeler Asay, Bumsoo Lee, Detlof von Winterfeldt. 2011. Exploring Reductions in London Underground Passenger Journeys Following the July 2005 Bombings. *Risk Analysis* no-no. [CrossRef]
- 303. Shinichi Doi, Keiji Yamada. 2010. Symbiotic technology for creating social innovation 30 years in the future. *AI & SOCIETY*. [CrossRef]
- 304. D. Wade Hands. 2010. BACK TO THE ORDINALIST REVOLUTION: BEHAVIORAL ECONOMIC CONCERNS IN EARLY MODERN CONSUMER CHOICE THEORY. *Metroeconomica* no-no. [CrossRef]
- 305. Shu-Heng Chen, Shu G. Wang. 2010. EMERGENT COMPLEXITY IN AGENT-BASED COMPUTATIONAL ECONOMICS. *Journal of Economic Surveys* no-no. [CrossRef]
- 306. K. Huijps, H. Hogeveen, G. Antonides, N. I. Valeeva, T. J. G. M. Lam, A. G. J. M. Oude Lansink. 2010. Sub-optimal economic behaviour with respect to mastitis management. *European Review of Agricultural Economics* 37:4, 553-568. [CrossRef]
- 307. Dan M. Kahan, Donald Braman, Geoffrey L. Cohen, John Gastil, Paul Slovic. 2010. Who Fears the HPV Vaccine, Who Doesn't, and Why? An Experimental Study of the Mechanisms of Cultural Cognition. Law and Human Behavior 34:6, 501-516. [CrossRef]
- 308. David A. Lake. 2010. Two Cheers for Bargaining Theory: Assessing Rationalist Explanations of the Iraq War. *International Security* 35:3, 7-52. [CrossRef]

- 309. Christine Clavien, Rebekka A. Klein. 2010. EAGER FOR FAIRNESS OR FOR REVENGE? PSYCHOLOGICAL ALTRUISM IN ECONOMICS. *Economics and Philosophy* 26:03, 267-290. [CrossRef]
- 310. William P. Bottom. 2010. Essence of Negotiation: Understanding Appeasement and "The Great Munich Stereotype". Negotiation Journal 26:4, 379-415. [CrossRef]
- 311. MIRA FARKA. 2010. THE ASYMMETRIC IMPACT OF "INFORMATIVE" AND "UNINFORMATIVE" FEDERAL OPEN MARKET COMMITTEE STATEMENTS ON ASSET PRICES. Contemporary Economic Policy no-no. [CrossRef]
- 312. JIŘÍ KABELE. 2010. The Agency/Structure Dilemma: A Coordination Solution. Journal for the Theory of Social Behaviour 40:3, 314-338. [CrossRef]
- 313. Jay P. Shimshack, Michael B. Ward. 2010. Mercury advisories and household health trade-offs. *Journal of Health Economics* 29:5, 674-685. [CrossRef]
- 314. Anne K. Soutter, Alison Gilmore, Billy O'Steen. 2010. How do High School Youths' Educational Experiences Relate to Well-Being? Towards a Trans-Disciplinary Conceptualization. *Journal of Happiness Studies*. [CrossRef]
- 315. Marta Gozzi, Paolo Cherubini, Costanza Papagno, Emanuela Bricolo. 2010. Recruitment of intuitive versus analytic thinking strategies affects the role of working memory in a gambling task. *Psychological Research*. [CrossRef]
- 316. Jeffrey M. Brunstrom, Jane Collingwood, Peter J. Rogers. 2010. Perceived volume, expected satiation, and the energy content of self-selected meals#. *Appetite* 55:1, 25-29. [CrossRef]
- 317. Keith T. Thomas, Allan D. Walker. 2010. The sharp end: Real life challenges in a complex activity space. *Journal of Public Affairs* 10:3, 186-199. [CrossRef]
- 318. Robert P. Hamlin. 2010. Cue-Based Decision Making. A new framework for understanding the uninvolved food consumer. *Appetite* 55:1, 89-98. [CrossRef]
- 319. SULEYMAN BASAK, HONGJUN YAN. 2010. Equilibrium Asset Prices and Investor Behaviour in the Presence of Money Illusion. *Review of Economic Studies* **77**:3, 914-936. [CrossRef]
- 320. Ika Darnhofer, Stéphane Bellon, Benoît Dedieu, Rebecka Milestad. 2010. Adaptiveness to enhance the sustainability of farming systems. A review. *Agronomy for Sustainable Development* 30:3, 545-555. [CrossRef]
- 321. O. Budzinski, I. Ruhmer. 2010. MERGER SIMULATION IN COMPETITION POLICY: A SURVEY. Journal of Competition Law and Economics 6:2, 277-319. [CrossRef]
- 322. Gordon L. Clark. 2010. HUMAN NATURE, THE ENVIRONMENT, AND BEHAVIOUR: EXPLAINING THE SCOPE AND GEOGRAPHICAL SCALE OF FINANCIAL DECISION-MAKING. *Geografiska Annaler: Series B, Human Geography* **92**:2, 159-173. [CrossRef]
- 323. Ruth E. Wachtel, Franklin Dexter. 2010. Review of Behavioral Operations Experimental Studies of Newsvendor Problems for Operating Room Management. Anesthesia & Analgesia 110:6, 1698-1710. [CrossRef]
- 324. Piers Steel. 2010. Arousal, avoidant and decisional procrastinators: Do they exist?. *Personality and Individual Differences* 48:8, 926-934. [CrossRef]
- 325. Julie A. Nelson. 2010. Getting past "rational man/emotional woman": comments on research programs in happiness economics and interpersonal relations. *International Review of Economics* 57:2, 233-253. [CrossRef]
- 326. D. Wade Hands. 2010. Stabilizing consumer choice: the role of 'true dynamic stability' and related concepts in the history of consumer choice theory. *The European Journal of the History of Economic Thought* 17:2, 313-343. [CrossRef]

- 327. Jef Vuchelen, Stijn Caekelbergh. 2010. Explaining public investment in Western Europe. *Applied Economics* 42:14, 1783-1796. [CrossRef]
- 328. Utpal M. Dholakia, Barbara E. Kahn, Randy Reeves, Aric Rindfleisch, David Stewart, Earl Taylor. 2010. Consumer Behavior in a Multichannel, Multimedia Retailing Environment. *Journal of Interactive Marketing* 24:2, 86-95. [CrossRef]
- 329. Archie Zariski. 2010. A Theory Matrix for Mediators. Negotiation Journal 26:2, 203-235. [CrossRef]
- 330. Stefan Goetze, Berthold Rittberger. 2010. A matter of habit? The sociological foundations of empowering the European Parliament. *Comparative European Politics* 8:1, 37-54. [CrossRef]
- 331. A. Etzioni, M. J. Piore, W. Streeck. 2010. Behavioural economics. Socio-Economic Review 8:2, 377-397. [CrossRef]
- 332. Jerome Feldman. 2010. Ecological expected utility and the mythical neural code. *Cognitive Neurodynamics* 4:1, 25-35. [CrossRef]
- 333. Victor K. Y. Chan. 2010. Using Neural Networks to Model the Behavior and Decisions of Gamblers, in Particular, Cyber-Gamblers. *Journal of Gambling Studies* 26:1, 35-52. [CrossRef]
- 334. Kirsten Bregn. 2010. The Logic of the New Pay Systems Revisited-in the Light of Experimental and Behavioral Economics. *International Journal of Public Administration* 33:4, 161-168. [CrossRef]
- 335. Julian N. Marewski, Lael J. Schooler, Gerd Gigerenzer. 2010. Five Principles for Studying People's Use of Heuristics. Acta Psychologica Sinica 42:1, 72-87. [CrossRef]
- 336. M. Baddeley. 2010. Herding, social influence and economic decision-making: socio-psychological and neuroscientific analyses. *Philosophical Transactions of the Royal Society B: Biological Sciences* 365:1538, 281-290. [CrossRef]
- 337. Yves Wautelet, Christophe Schinckus, Manuel Kolp. 2010. Towards Knowledge Evolution in Software Engineering. International Journal of Information Technologies and Systems Approach 3:1, 21-40. [CrossRef]
- 338. Brian K. Burton, Michael G. Goldsby. 2010. The Moral Floor: A Philosophical Examination of the Connection Between Ethics and Business. *Journal of Business Ethics* **91**:1, 145-154. [CrossRef]
- 339. Fabian Gouret, Guillaume Hollard. 2010. When Kahneman meets Manski: Using dual systems of reasoning to interpret subjective expectations of equity returns. *Journal of Applied Econometrics* n/an/a. [CrossRef]
- 340. Bibliography 163-200. [CrossRef]
- 341. Josef Kaenzig, Rolf Wüstenhagen. 2010. The Effect of Life Cycle Cost Information on Consumer Investment Decisions Regarding Eco-Innovation. *Journal of Industrial Ecology* 14:1, 121-136. [CrossRef]
- 342. Alok Kumar. 2009. Hard-to-Value Stocks, Behavioral Biases, and Informed Trading. *Journal of Financial and Quantitative Analysis* 44:06, 1375. [CrossRef]
- 343. Kristen Renwick Monroe, William Chiu, Adam Martin, Bridgette Portman. 2009. What Is Political Psychology?. *Perspectives on Politics* 7:04, 859. [CrossRef]
- 344. Randall G. Holcombe. 2009. The behavioral foundations of Austrian economics. *The Review of Austrian Economics* 22:4, 301-313. [CrossRef]
- 345. VICTOR STANGO, JONATHAN ZINMAN. 2009. Exponential Growth Bias and Household Finance. *The Journal of Finance* 64:6, 2807-2849. [CrossRef]
- 346. Jianbiao Li, Guangqian Ren, Qiuhua Ma, Lüke Liu. 2009. An experimental study on investors' status quo bias and its determinants. *Frontiers of Business Research in China* **3**:4, 543-565. [CrossRef]
- 347. Constant D. Beugré. 2009. Exploring the neural basis of fairness: A model of neuro-organizational justice. *Organizational Behavior and Human Decision Processes* 110:2, 129-139. [CrossRef]

- 348. S. C. Edwards, S. C. Pratt. 2009. Rationality in collective decision-making by ant colonies. Proceedings of the Royal Society B: Biological Sciences 276:1673, 3655-3661. [CrossRef]
- 349. W. J. W. Botzen, J. C. J. H. Aerts, J. C. J. M. van den Bergh. 2009. Dependence of flood risk perceptions on socioeconomic and objective risk factors. *Water Resources Research* 45:10, n/a-n/a. [CrossRef]
- 350. Gordon L. Clark, Emiko Caerlewy-Smith, John C. Marshall. 2009. Solutions to the Asset Allocation Problem by Informed Respondents: The Significance of the Size-of-Bet and the 1/N Heuristic. *Risk Management and Insurance Review* 12:2, 251-271. [CrossRef]
- 351. Agnès Festré, Pierre Garrouste. 2009. The economic analysis of social norms: A reappraisal of Hayek's legacy. *The Review of Austrian Economics* 22:3, 259-279. [CrossRef]
- 352. Michael Schwarz. 2009. Is Psychology Based on a Methodological Error?. *Integrative Psychological and Behavioral Science* **43**:3, 185-213. [CrossRef]
- 353. Randall G. Holcombe. 2009. A reformulation of the foundations of welfare economics. *The Review of Austrian Economics* 22:3, 209-224. [CrossRef]
- 354. Ji-Ping YANG, Jian-Jun ZHENG. 2009. The Effect of Emotion on the Quality of Crisis Decisionmaking. Acta Psychologica Sinica 41:6, 481-491. [CrossRef]
- 355. Donald R. Kinder, Katherine W. Drake. 2009. Myrdal's Prediction. *Political Psychology* 30:4, 539-568. [CrossRef]
- 356. Fintan J. Costello. 2009. How probability theory explains the conjunction fallacy. *Journal of Behavioral Decision Making* 22:3, 213-234. [CrossRef]
- 357. Ray Paternoster, Greg Pogarsky. 2009. Rational Choice, Agency and Thoughtfully Reflective Decision Making: The Short and Long-Term Consequences of Making Good Choices. *Journal of Quantitative Criminology* 25:2, 103-127. [CrossRef]
- 358. Tim Schwanen, Dick Ettema. 2009. Coping with unreliable transportation when collecting children: Examining parents' behavior with cumulative prospect theory. *Transportation Research Part A: Policy and Practice* **43**:5, 511-525. [CrossRef]
- 359. Dejan Trifunović, Bojan Ristić, Marko Ivković, Svetozar Tanasković, Leo Italiano, Stefania Tattoni. 2009. FDI's Impact on Transitional Countries, Serbia as a Rational Choice: The FIAT-ZASTAVA Case. Transition Studies Review 16:2, 269-286. [CrossRef]
- 360. Anastasiya Pocheptsova, On Amir, Ravi Dhar, Roy F Baumeister. 2009. Deciding Without Resources: Resource Depletion and Choice in Context. *Journal of Marketing Research* 46:3, 344-355. [CrossRef]
- 361. Rob Aalbers, Eline van der Heijden, Jan Potters, Daan van Soest, Herman Vollebergh. 2009. Technology adoption subsidies: An experiment with managers#. *Energy Economics* 31:3, 431-442. [CrossRef]
- 362. Caspar G. Chorus, Theo A. Arentze, Harry J.P. Timmermans. 2009. Traveler compliance with advice: A Bayesian utilitarian perspective. *Transportation Research Part E: Logistics and Transportation Review* 45:3, 486-500. [CrossRef]
- 363. Yuki Sampei, Midori Aoyagi-Usui. 2009. Mass-media coverage, its influence on public awareness of climate-change issues, and implications for Japan's national campaign to reduce greenhouse gas emissions. *Global Environmental Change* 19:2, 203-212. [CrossRef]
- 364. Henk Folmer. 2009. Why Sociology is Better Conditioned to Explain Economic Behaviour than Economics. *Kyklos* 62:2, 258-274. [CrossRef]
- 365. GONUL DOĞAN, MARCEL ASSEN. 2009. Testing Models of Pure Exchange. The Journal of Mathematical Sociology 33:2, 97-128. [CrossRef]
- 366. Jifeng Mu, Gang Peng, Douglas L. MacLachlan. 2009. Effect of risk management strategy on NPD performance. *Technovation* 29:3, 170-180. [CrossRef]

- 367. Germano Resconi, Boris Kovalerchuk. 2009. Agents' model of uncertainty. *Knowledge and Information Systems* 18:2, 213-229. [CrossRef]
- 368. Gordon L Clark, Pierre Pestieau. 2009. The 'new' paternalism, consultation and consent: Expectations of UK participants in defined contribution and self-directed retirement savings schemes. *Pensions: An International Journal* 14:1, 58-74. [CrossRef]
- 369. C DEFEUILLEY. 2009. Retail competition in electricity markets. *Energy Policy* 37:2, 377-386. [CrossRef]
- 370. Cars Hommes, Florian WagenerComplex Evolutionary Systems in Behavioral Finance 217-276. [CrossRef]
- 371. June Carbone, Naomi CahnResearch in Law and Economics 24, . [CrossRef]
- 372. Lennart Sjoberg, Elisabeth Engelberg. 2009. Attitudes to Economic Risk Taking, Sensation Seeking and Values of Business Students Specializing in Finance. *Journal of Behavioral Finance* 10:1, 33-43. [CrossRef]
- 373. Keis Ohtsuka, Chi Chuen Chan. 2009. Desperate Housewives: An Analysis of the Characterisations of Female Gamblers Portrayed in Gambling Movies in Hong Kong. *International Journal of Mental Health and Addiction* 7:1, 229-238. [CrossRef]
- 374. Daniel KahnemanRemarks on Neuroeconomics 523-526. [CrossRef]
- 375. Brian Sternthal, Andrea Bonezzi. 2009. Consumer decision making and aging: A commentary. *Journal of Consumer Psychology* 19:1, 23-27. [CrossRef]
- 376. Michael Bar-Eli, Ofer H. Azar, Yotam Lurie(Ir)rationality in action: do soccer players and goalkeepers fail to learn how to best perform during a penalty kick? 97-108. [CrossRef]
- 377. Antonio DamasioNeuroscience and the Emergence of Neuroeconomics 207-213. [CrossRef]
- 378. Mark Schenkel, Charles Matthews, Matthew Ford. 2009. Making rational use of 'irrationality'? Exploring the role of need for cognitive closure in nascent entrepreneurial activity. *Entrepreneurship* & Regional Development 21:1, 51-76. [CrossRef]
- 379. M. Karl Healey, Lynn Hasher. 2009. Limitations to the deficit attenuation hypothesis: Aging and decision making. *Journal of Consumer Psychology* **19**:1, 17-22. [CrossRef]
- 380. Shinji Teraji. 2009. The economics of possible selves. *The Journal of Socio-Economics* 38:1, 45-51. [CrossRef]
- Tjaco H Walvis. 2008. Three laws of branding: Neuroscientific foundations of effective brand building. *Journal of Brand Management* 16:3, 176-194. [CrossRef]
- 382. stephen graham saunders. 2008. TOWARD BRIDGING THE GAP BETWEEN THEORY AND EMPIRICAL REALITY. *South African Journal of Economics* **76**:4, 738-748. [CrossRef]
- 383. Steven S. Wildman. 2008. Communication and Economics: Two Imperial Disciplines and Too Little Collaboration. *Journal of Communication* 58:4, 693-706. [CrossRef]
- 384. Alessandro Lanteri, Chiara Chelini, Salvatore Rizzello. 2008. An Experimental Investigation of Emotions and Reasoning in the Trolley Problem. *Journal of Business Ethics* 83:4, 789-804. [CrossRef]
- 385. Xinxin Li, Lorin M. Hitt. 2008. Self-Selection and Information Role of Online Product Reviews. Information Systems Research 19:4, 456-474. [CrossRef]
- 386. SAULO D. BARBOSA, JILL KICKUL, BRETT R. SMITH. 2008. THE ROAD LESS INTENDED: INTEGRATING ENTREPRENEURIAL COGNITION AND RISK IN ENTREPRENEURSHIP EDUCATION. Journal of Enterprising Culture 16:04, 411-439. [CrossRef]
- 387. Jakob Haan. 2008. The effect of ECB communication on interest rates: An assessment. *The Review of International Organizations* 3:4, 375-398. [CrossRef]

- 388. Nicholas Barberis, Ming Huang. 2008. Stocks as Lotteries: The Implications of Probability Weighting for Security Prices. *American Economic Review* 98:5, 2066-2100. [Abstract] [View PDF article] [PDF with links]
- 389. Alan S. Blinder, Michael Ehrmann, Marcel Fratzscher, Jakob De Haan, David-Jan Jansen. 2008. Central Bank Communication and Monetary Policy: A Survey of Theory and Evidence. *Journal of Economic Literature* 46:4, 910-945. [Abstract] [View PDF article] [PDF with links]
- 390. K GODA, H HONG. 2008. Application of cumulative prospect theory: Implied seismic design preference. *Structural Safety* **30**:6, 506-516. [CrossRef]
- 391. David Hirshleifer. 2008. Psychological Bias as a Driver of Financial Regulation. *European Financial Management* 14:5, 856-874. [CrossRef]
- 392. L. Venkatachalam. 2008. Behavioral economics for environmental policy. *Ecological Economics* 67:4, 640-645. [CrossRef]
- 393. C. Leigh Anderson, Alison Cullen, Kostas Stamoulis. 2008. Preference variability along the policy chain in Vietnam. *The Journal of Socio-Economics* **37**:5, 1729-1745. [CrossRef]
- 394. Antony Millner. 2008. Getting the Most out of Ensemble Forecasts: A Valuation Model Based on User–Forecast Interactions. *Journal of Applied Meteorology and Climatology* 47:10, 2561-2571. [CrossRef]
- 395. Yi-Tsung Lee, Yu-Jane Liu, Ning Zhu. 2008. The Costs of Owning Employer Stocks: Lessons from Taiwan. *Journal of Financial and Quantitative Analysis* **43**:03, 717. [CrossRef]
- 396. Don P. Clausing, Konstantinos V. Katsikopoulos. 2008. Rationality in systems engineering: Beyond calculation or political action. *Systems Engineering* 11:4, 309-328. [CrossRef]
- 397. F. Gino, G. Pisano. 2008. Toward a Theory of Behavioral Operations. *Manufacturing & Service Operations Management* 10:4, 676-691. [CrossRef]
- 398. Franziska Rischkowsky, Thomas Döring. 2008. Consumer Policy in a Market Economy Considerations from the Perspective of the Economics of Information, the New Institutional Economics as well as Behavioural Economics. *Journal of Consumer Policy* 31:3, 285-313. [CrossRef]
- Hans Mathias Thjømøe. 2008. Branding, cheating the customer and other heretical thoughts. *Journal of Brand Management* 16:1-2, 105-109. [CrossRef]
- 400. Werner Neu. 2008. Making economic sense of brain models: a survey and interpretation of the literature. *Journal of Bioeconomics* 10:2, 165-192. [CrossRef]
- 401. GORDON L. CLARK, KENDRA STRAUSS. 2008. Individual pension-related risk propensities: the effects of socio-demographic characteristics and a spousal pension entitlement on risk attitudes. *Ageing & Society* 28:06. [CrossRef]
- 402. A PERRELS. 2008. Wavering between radical and realistic sustainable consumption policies: in search for the best feasible trajectories#. *Journal of Cleaner Production* **16**:11, 1203-1217. [CrossRef]
- 403. I HOZO, M SCHELL, B DJULBEGOVIC. 2008. Decision-Making When Data and Inferences Are Not Conclusive: Risk-Benefit and Acceptable Regret Approach. *Seminars in Hematology* 45:3, 150-159. [CrossRef]
- 404. Alok Kumar, Sonya Seongyeon Lim. 2008. How Do Decision Frames Influence the Stock Investment Choices of Individual Investors?. *Management Science* **54**:6, 1052-1064. [CrossRef]
- 405. Ravi Kanbur, Jukka Pirttilä, Matti Tuomala. 2008. Moral Hazard, Income Taxation and Prospect Theory*. *Scandinavian Journal of Economics* 110:2, 321-337. [CrossRef]
- 406. J GODEK, K MURRAY. 2008. Willingness to pay for advice: The role of rational and experiential processing#. Organizational Behavior and Human Decision Processes 106:1, 77-87. [CrossRef]

- 407. Philipp Broeckelmann, Andrea Groeppel-Klein. 2008. Usage of mobile price comparison sites at the point of sale and its influence on consumers' shopping behaviour. *The International Review of Retail, Distribution and Consumer Research* 18:2, 149-166. [CrossRef]
- 408. Tim Schwanen. 2008. Struggling with Time: Investigating Coupling Constraints. *Transport Reviews* 28:3, 337-356. [CrossRef]
- 409. Carlo Reverberi, Patrice Rusconi, Eraldo Paulesu, Paolo Cherubini. 2008. Response demands and the recruitment of heuristic strategies in syllogistic reasoning. *The Quarterly Journal of Experimental Psychology* **62**:3, 513-530. [CrossRef]
- 410. E. Yujuico. 2008. Connecting the dots in social entrepreneurship through the capabilities approach. *Socio-Economic Review* 6:3, 493-513. [CrossRef]
- 411. S. Dow. 2008. Mainstream Methodology, Financial Markets and Global Political Economy. Contributions to Political Economy 27:1, 13-29. [CrossRef]
- 412. Henrik Andersson, Mikael Svensson. 2008. Cognitive ability and scale bias in the contingent valuation method. *Environmental and Resource Economics* **39**:4, 481-495. [CrossRef]
- 413. Austin B. Frakt, Steven D. Pizer. 2008. Attribute substitution in early enrollment decisions into Medicare prescription drug plans. *Health Economics* 17:4, 513-521. [CrossRef]
- 414. K. Goda, H. P. Hong. 2008. Implied Preference for Seismic Design Level and Earthquake Insurance. *Risk Analysis* 28:2, 523-537. [CrossRef]
- 415. Adi Livnat, Nicholas Pippenger. 2008. Systematic mistakes are likely in bounded optimal decisionmaking systems. *Journal of Theoretical Biology* **250**:3, 410-423. [CrossRef]
- 416. Karsten Mause. 2008. The Tragedy of the Commune: Learning from worst-case scenarios. *The Journal of Socio-Economics* 37:1, 308-327. [CrossRef]
- 417. Gordon L. Clark, Roger Urwin. 2008. Making Pension Boards Work: The Critical Role of Leadership. *Rotman International Journal of Pension Management* 1:1, 38-45. [CrossRef]
- 418. Alan Schwartz. 2008. How Much Irrationality Does the Market Permit?. *The Journal of Legal Studies* **37**:1, 131-159. [CrossRef]
- 419. N BARBERISThe Loss Aversion/Narrow Framing Approach to the Equity Premium Puzzle 199-229. [CrossRef]
- 420. K L BLACKSTOCK, C E CARTER. 2007. Operationalising sustainability science for a sustainability directive? Reflecting on three pilot projects. *The Geographical Journal* **173**:4, 343-357. [CrossRef]
- 421. Thomas A. Faunce. 2007. Nanotechnology in Global Medicine and Human Biosecurity: Private Interests, Policy Dilemmas, and the Calibration of Public Health Law. *The Journal of Law, Medicine* & Ethics 35:4, 629-642. [CrossRef]
- 422. Michael A. Salinger, Pauline M. Ippolito, Joel L. Schrag. 2007. Economics at the FTC: Pharmaceutical Patent Dispute Settlements and Behavioral Economics. *Review of Industrial Organization* **31**:2, 85-105. [CrossRef]
- 423. Alexandre Linhares, Paulo Brum. 2007. Understanding Our Understanding of Strategic Scenarios: What Role Do Chunks Play?. *Cognitive Science* **31**:6, 989-1007. [CrossRef]
- 424. Charlie Wilson, Hadi Dowlatabadi. 2007. Models of Decision Making and Residential Energy Use. Annual Review of Environment and Resources 32:1, 169-203. [CrossRef]
- 425. Jack J. Vromen. 2007. Neuroeconomics as a Natural Extension of Bioeconomics: The Shifting Scope of Standard Economic Theory. *Journal of Bioeconomics* **9**:2, 145-167. [CrossRef]
- 426. Gordon L Clark, Janelle Knox-Hayes. 2007. Mapping UK pension benefits and the intended purchase of annuities in the aftermath of the 1990s stock market bubble. *Transactions of the Institute of British Geographers* 32:4, 539-555. [CrossRef]

- 427. DIETER GERDESMEIER, FRANCESCO PAOLO MONGELLI, BARBARA ROFFIA. 2007. The Eurosystem, the U.S. Federal Reserve, and the Bank of Japan: Similarities and Differences. *Journal of Money, Credit and Banking* 39:7, 1785-1819. [CrossRef]
- 428. A. I. Houston, J. M. McNamara, M. D. Steer. 2007. Do we expect natural selection to produce rational behaviour?. *Philosophical Transactions of the Royal Society B: Biological Sciences* 362:1485, 1531-1543. [CrossRef]
- 429. R. Bogacz, M. Usher, J. Zhang, J. L. McClelland. 2007. Extending a biologically inspired model of choice: multi-alternatives, nonlinearity and value-based multidimensional choice. *Philosophical Transactions of the Royal Society B: Biological Sciences* 362:1485, 1655-1670. [CrossRef]
- 430. C. Perrings. 2007. Future challenges. Proceedings of the National Academy of Sciences 104:39, 15179-15180. [CrossRef]
- 431. Marie-Laure Baron. 2007. Defining the frontiers of the firm through property rights allocation: The case of the French retailer cooperative Leclerc. *Review of Social Economy* **65**:3, 293-317. [CrossRef]
- 432. Botond Kőszegi, Matthew Rabin. 2007. Reference-Dependent Risk Attitudes. *American Economic Review* 97:4, 1047-1073. [Abstract] [View PDF article] [PDF with links]
- 433. Michael Schaefer, Michael Rotte. 2007. Thinking on luxury or pragmatic brand products: Brain responses to different categories of culturally based brands. *Brain Research* 1165, 98-104. [CrossRef]
- 434. J GOWDY, R JULIA. 2007. Technology and petroleum exhaustion: Evidence from two megaoilfields. *Energy* 32:8, 1448-1454. [CrossRef]
- 435. John A. Doukas, Dimitris Petmezas. 2007. Acquisitions, Overconfident Managers and Self-attribution Bias. *European Financial Management* 13:3, 531-577. [CrossRef]
- 436. Amos Schurr, Ido Erev. 2007. The effect of base rate, careful analysis, and the distinction between decisions from experience and from description. *Behavioral and Brain Sciences* **30**:03. [CrossRef]
- 437. Robert R. Prechter, Wayne D. Parker. 2007. The Financial/Economic Dichotomy in Social Behavioral Dynamics: The Socionomic Perspective. *Journal of Behavioral Finance* **8**:2, 84-108. [CrossRef]
- 438. Till Grüne-Yanoff. 2007. Bounded Rationality. Philosophy Compass 2:3, 534-563. [CrossRef]
- 439. Frank J. Fabozzi, Sergio Focardi, Caroline Jonas. 2007. Trends in quantitative equity management: survey results. *Quantitative Finance* 7:2, 115-122. [CrossRef]
- 440. Randall G. Holcombe, Lawrence W. Kenny. 2007. Evidence on voter preferences from unrestricted choice referendums. *Public Choice* 131:1-2, 197-215. [CrossRef]
- 441. John Gowdy. 2007. Can Economic Theory Stop being a Cheerleader for Corporate Capitalism?. *Psychological Inquiry* 18:1, 33-35. [CrossRef]
- 442. MICHAEL EHRMANN, MARCEL FRATZSCHER. 2007. Communication by Central Bank Committee Members: Different Strategies, Same Effectiveness?. *Journal of Money, Credit and Banking* 39:2-3, 509-541. [CrossRef]
- 443. Boris M. Velichkovsky. 2007. Towards an Evolutionary Framework for Human Cognitive Neuroscience. *Biological Theory* 2:1, 3-6. [CrossRef]
- 444. Werner Callebaut. 2007. Herbert Simon's Silent Revolution. Biological Theory 2:1, 76-86. [CrossRef]
- 445. Amrit Tiwana, Jijie Wang, Mark Keil, Punit Ahluwalia. 2007. The Bounded Rationality Bias in Managerial Valuation of Real Options: Theory and Evidence from IT Projects. *Decision Sciences* 38:1, 157-181. [CrossRef]
- 446. Howard Moskowitz, Samuel Rabino, Alex Gofman, Daniel Moskowitz. 2007. Effective and confident communications in the midst of a major crisis: An experiment in the pharmaceutical context. *International Journal of Pharmaceutical and Healthcare Marketing* 1:4, 318-348. [CrossRef]

- 447. John Urry, Robert Dingwall, Ian Gough, Paul Ormerod, Doreen Massey, John Scott, Nigel Thrift. 2007. What is 'social' about social science?. *Twenty-First Century Society* **2**:1, 95-119. [CrossRef]
- 448. Irene Van Staveren. 2007. Beyond Utilitarianism and Deontology: Ethics in Economics. *Review of Political Economy* 19:1, 21-35. [CrossRef]
- 449. Craig R. Carter, Lutz Kaufmann, Alex Michel. 2007. Behavioral supply management: a taxonomy of judgment and decision-making biases. *International Journal of Physical Distribution & Logistics Management* **37**:8, 631-669. [CrossRef]
- 450. Daphne R. Raban. 2007. User-centered evaluation of information: a research challenge. *Internet Research* 17:3, 306-322. [CrossRef]
- 451. Benjamin Djulbegovic, Gary H Lyman. 2006. Screening mammography at 40–49 years: regret or no regret?. *The Lancet* **368**:9552, 2035-2037. [CrossRef]
- 452. Ulf Böckenholt. 2006. Thurstonian-Based Analyses: Past, Present, and Future Utilities. *Psychometrika* 71:4, 615-629. [CrossRef]
- 453. Volkan Topalli. 2006. The Seductive Nature of Autotelic Crime: How Neutralization Theory Serves as a Boundary Condition for Understanding Hardcore Street Offending. *Sociological Inquiry* **76**:4, 475-501. [CrossRef]
- 454. Wuyang Hu, Wiktor L. Adamowicz, Michele M. Veeman. 2006. Labeling Context and Reference Point Effects in Models of Food Attribute Demand. *American Journal of Agricultural Economics* 88:4, 1034-1049. [CrossRef]
- 455. Robin M. Hogarth, Natalia Karelaia. 2006. Regions of Rationality: Maps for Bounded Agents. *Decision Analysis* 3:3, 124-144. [CrossRef]
- 456. Ikrom Artikov, Stacey J. Hoffman, Gary D. Lynne, Lisa M. Pytlik Zillig, Qi Hu, Alan J. Tomkins, Kenneth G. Hubbard, Michael J. Hayes, William Waltman. 2006. Understanding the Influence of Climate Forecasts on Farmer Decisions as Planned Behavior*. *Journal of Applied Meteorology and Climatology* 45:9, 1202-1214. [CrossRef]
- 457. Eldad Yechiam, David V. Budescu. 2006. The Sensitivity of Probability Assessments to Time Units and Performer Characteristics. *Decision Analysis* 3:3, 177-193. [CrossRef]
- 458. Frederick J. Zimmerman. 2006. Agreeing on more than chicken soup: Intra-household decisionmaking and treatment for child psychopathology. *Review of Economics of the Household* 4:3, 229-252. [CrossRef]
- 459. Nicholas Barberis, Ming Huang, Richard H. Thaler. 2006. Individual Preferences, Monetary Gambles, and Stock Market Participation: A Case for Narrow Framing. *American Economic Review* 96:4, 1069-1090. [Abstract] [View PDF article] [PDF with links]
- 460. Gary D. Lynne. 2006. Toward a dual motive metaeconomic theory. *The Journal of Socio-Economics* 35:4, 634-651. [CrossRef]
- 461. Oran R. Young, Michael Zürn. 2006. The International Regimes Database: Designing and Using a Sophisticated Tool for Institutional Analysis. *Global Environmental Politics* **6**:3, 121-143. [CrossRef]
- 462. John Hobcraft. 2006. The ABC of demographic behaviour: How the interplays of alleles, brains, and contexts over the life course should shape research aimed at understanding population processes. *Population Studies* **60**:2, 153-187. [CrossRef]
- 463. Ariel Rubinstein. 2006. Dilemmas of an Economic Theorist. Econometrica 74:4, 865-883. [CrossRef]
- 464. Michael Schaefer, Harald Berens, Hans-Jochen Heinze, Michael Rotte. 2006. Neural correlates of culturally familiar brands of car manufacturers. *NeuroImage* **31**:2, 861-865. [CrossRef]
- 465. Yves Alarie, Georges Dionne. 2006. Lottery qualities. Journal of Risk and Uncertainty 32:3, 195-216. [CrossRef]

- 466. GUR HUBERMAN, WEI JIANG. 2006. Offering versus Choice in 401(k) Plans: Equity Exposure and Number of Funds. *The Journal of Finance* 61:2, 763-801. [CrossRef]
- 467. Petter Holme, Gourab Ghoshal. 2006. Dynamics of Networking Agents Competing for High Centrality and Low Degree. *Physical Review Letters* **96**:9. . [CrossRef]
- 468. D CHAKRAVARTI. 2006. Voices Unheard: The Psychology of Consumption in Poverty and Development. *Journal of Consumer Psychology* 16:4, 363-376. [CrossRef]
- 469. Thomas BrennerChapter 18 Agent Learning Representation: Advice on Modelling Economic Learning 895-947. [CrossRef]
- 470. Michael Useem, Andy Zelleke. 2006. Oversight and Delegation in Corporate Governance: deciding what the board should decide. *Corporate Governance: An International Review* 14:1, 2-12. [CrossRef]
- 471. Cars H. HommesChapter 23 Heterogeneous Agent Models in Economics and Finance 1109-1186. [CrossRef]
- 472. Marie-Hélène Broihanne, Maxime Merli, Patrick Roger. 2006. Théorie comportementale du portefeuille. *Revue économique* 57:2, 297. [CrossRef]
- 473. Ulf Böckenholt, Rung-Ching Tsail4 Random-Effects Models for Preference Data 447-468. [CrossRef]
- 474. Nannerl O. Keohane. 2005. On Leadership. Perspectives on Politics 3:04. . [CrossRef]
- 475. Otmar Issing. 2005. Kommunikation, Transparenz, Rechenschaft Geldpolitik im 21. Jahrhundert. *Perspektiven der Wirtschaftspolitik* 6:4, 521-540. [CrossRef]
- 476. Ulf BöckenholtScaling of Preferential Choice . [CrossRef]
- 477. Marcel Fafchamps, Ruth Vargas Hill. 2005. Selling at the Farmgate or Traveling to Market. *American Journal of Agricultural Economics* 87:3, 717-734. [CrossRef]
- 478. Randall G. Holcombe. 2005. Government growth in the twenty-first century. *Public Choice* 124:1-2, 95-114. [CrossRef]
- 479. Y FU. 2005. Measuring personal networks with daily contacts: a single-item survey question and the contact diary. *Social Networks* 27:3, 169-186. [CrossRef]
- 480. J GOWDY, J ERICKSON. 2005. Ecological economics at a crossroads. *Ecological Economics* 53:1, 17-20. [CrossRef]
- 481. Colin Camerer, George Loewenstein, Drazen Prelec. 2005. Neuroeconomics: How Neuroscience Can Inform Economics. *Journal of Economic Literature* 43:1, 9-64. [Abstract] [View PDF article] [PDF with links]
- 482. Albert Maydeu-Olivares, Ulf Böckenholt. 2005. Structural Equation Modeling of Paired-Comparison and Ranking Data. *Psychological Methods* **10**:3, 285-304. [CrossRef]
- 483. Roger W. Mills. 2005. Assessing Growth Estimates in IPO Valuations-A Case Study. *Journal of Applied Corporate Finance* 17:1, 73-78. [CrossRef]
- 484. Robert M. Yarbrough. 2005. Teaching Bioeconomics. Journal of Bioeconomics 7:1, 1-38. [CrossRef]
- 485. Michael Mandler. 2004. Status quo maintenance reconsidered: changing or incomplete preferences?
 *. The Economic Journal 114:499, F518-F535. [CrossRef]
- 486. Ira Horowitz. 2004. Aggregating Expert Ratings Using Preference-Neutral Weights: The Case of the College Football Polls. *Interfaces* 34:4, 314-320. [CrossRef]
- 487. Michael Lokshin, Ruslan Yemtsov. 2004. Combining Longitudinal Household and Community Surveys for Evaluation of Social Transfers: infrastructure rehabilitation projects in rural Georgia. *Journal of Human Development* 5:2, 265-277. [CrossRef]
- 488. Ulf Böckenholt. 2004. Comparative Judgments as an Alternative to Ratings: Identifying the Scale Origin. *Psychological Methods* **9**:4, 453-465. [CrossRef]

- S. Nuri Erbas. 2004. Ambiguity, Transparency, and Institutional Strength. *IMF Working Papers* 04, 1. [CrossRef]
- 490. Steven M. Suranovic, Robert S. GoldfarbA Behavioral Model of Cyclical Dieting 49-78. [CrossRef]
- 491. Floris HeukelomBuilding and Defining Behavioral Economics 1-29. [CrossRef]
- 492. Heather Schofield, Sendhil Mullainathan The psychology of nutrition messages 145-172. [CrossRef]
- 493. Yves Wautelet, Christophe Schinckus, Manuel KolpTowards Knowledge Evolution in Software Engineering 8-24. [CrossRef]
- 494. Germano Resconi, Boris KovalerchukAgents in Quantum and Neural Uncertainty 50-77. [CrossRef]
- 495. Hamidreza Amouzegar, Mohammad Jafar Tarokh, Anahita Naghilouye HidajiA New SOA Security Model to Protect Against Web Competitive Intelligence Attacks by Software Agents 327-336. [CrossRef]
- 496. Brian J. LoasbyHAYEK'S THEORY OF THE MIND 101-134. [CrossRef]
- 497. Kathleen L. WolfCity Trees and Consumer Response in Retail Business Districts 152-172. [CrossRef]